

THE IMPACT OF BRAND AWARENESS, STORE IMAGE, AND SERVICE QUALITY IN PROMOTING BRAND EQUITY IN THE BANKING INDUSTRY OF SIERRA LEONE

MASTER THESIS

ABU BAKARR ALEX SESAY

Nicosia August, 2022

NEAR EAST UNIVERSITY INSTITUTE OF GRADUATE STUDIES DEPARTMENT OF BUSINESS ADMINISTRATION

THE IMPACT OF BRAND AWARENESS, STORE IMAGE, AND SERVICE QUALITY IN PROMOTING BRAND EQUITY IN THE BANKING INDUSTRY OF SIERRA LEONE

MASTER THESIS

ABU BAKARR ALEX SESAY

Supervisor Dr. Laith Tashtoush

> Nicosia August, 2022

Approval

We certify that we have read the thesis submitted by ABU BAKARR ALEX SESAY titled THE IMPACT OF BRAND AWARENESS, STORE IMAGE, AND SERVICE QUALITY IN PROMOTING BRAND EQUITY IN THE BANKING INDUSTRY OF SIERRA LEONE" and that in our combined opinion it is fully adequate, in scope and in quality, as a thesis for the degree of Master of Educational Sciences.

Examining Committee

Name-Surname

Signature

Head of the Committee: Dr. Ala Fathi Assi

Committee Member*: Assist. Prof. Dr. Ayşe Gözde Koyuncu

Supervisor:

Dr. Laith Tashtoush

Approved by the Head of the Department

Head of Department

Approved by the Institute of Graduate Studies



2

Declaration

I hereby declare that all information, documents, analysis and results in this thesis have been collected and presented according to the academic rules and ethical guidelines of Institute of Graduate Studies, Near East University. I also declare that as required by these rules and conduct, I have fully cited and referenced information and data that are not original to this study.

ABU BAKARR ALEX SESAY

/ /2022

Acknowledgments

Throughout the writing of this thesis, I have received a great deal of support and assistance.

I would first like to thank my supervisor, Dr. Laith Tashtoush, whose expertise was invaluable in formulating the research questions and methodology. Your insightful feedback pushed me to sharpen my thinking and brought my work to a higher level.

In addition, I would like to thank my parents for their wise counsel and sympathetic ear. You are always there for me. Finally, I could not have completed this thesis without the support of my friends who provided stimulating discussions as well as happy distractions to rest my mind outside of my research.

ABU BAKARR ALEX SESAY

Abstract

The Impact of Brand Awareness, Store Image, And Service Quality in Promoting Brand Equity in The Banking Industry of Sierra Leone

Alex Sesay, Abu Bakarr

Supervisor, Dr. Laith Tashtoush

MA, Department of Business Administration

August, 2022, 92 pages

The general objective of the study is to determine the impact of brand awareness, store image, and service quality in promoting brand equity for banking industry in Sierra Leone. The study adopted a quantitative researcher design. Data for the study were gathered from both primary and secondary sources. The study employed convenience sampling technique to select 450 respondents to administer the questionnaires. The data were analyzed using SPSS. The study discovered that an unprecedented majority of the respondents had adequate knowledge on brand awareness, store image, and service quality in promoting brand equity. Although some respondents acknowledged that branding is about logo, service quality, pricing, coverage image, signage and slogan. A whopping majority seem this as combination of all the above. The study had also discovered that branding has a significant association with branding and customer buying behavior. As a result, there has been an accompanying rapid development, and fierce competition in the market with a range of new financial service providers emerging with varied product, services and advertising messages targeted at retaining their existing customer base as well as targeting new and potential customers. Moreover, customer satisfaction is very vital to attract and retain customers especially in the banking business environment where competitions among firms are very intensive and unabated. Customer services such as complains handling, waiting time and customer relationship management must be given a maximum attention.

Keywords: brand awareness, store image, service quality brand equity, Sierra Leone

The Impact of Brand Awareness, Store Image, And Service Quality in Promoting Brand Equity in The Banking Industry of Sierra Leone

ÖZ

Alex Sesay, Abu Bakarr

Supervisor, Dr. Laith Tashtoush

MA, Department of Business Administration

August, 2022, 92 pages

Çalışmanın genel amacı, Sierra Leone'de bankacılık sektöründe marka değerini teşvik etmede marka bilinirliği, mağaza imajı ve hizmet kalitesinin etkisini belirlemektir. Çalışma, nicel bir araştırmacı tasarımı benimsemiştir. Araştırmanın verileri hem birincil hem de ikincil kaynaklardan toplanmıştır. Çalışma, anketleri uygulamak üzere 450 katılımcıyı seçmek için bir kolayda örnekleme tekniği kullanmıştır. Veriler SPSS kullanılarak analiz edildi. Araştırma, ankete katılanların emsalsiz bir çoğunluğunun marka denkliğini tesvik etmede marka bilinirliği, mağaza imajı ve hizmet kalitesi hakkında yeterli bilgiye sahip olduğunu keşfetti. Bazı katılımcılar marka bilinci oluşturmanın logo, hizmet kalitesi, fiyatlandırma, kapsam imajı, tabela ve slogan ile ilgili olduğunu kabul etse de. Büyük bir çoğunluk bunu yukarıdakilerin bir kombinasyonu olarak görüyor. Çalışma ayrıca markalaşmanın markalaşma ve müşteri satın alma davranışı ile önemli bir ilişkisi olduğunu keşfetti. Sonuç olarak, çeşitli ürünler, hizmetler ve mevcut müşteri tabanlarını korumayı hedefleyen reklam mesajları ile ortaya çıkan bir dizi yeni finansal hizmet sağlayıcı ile birlikte piyasada hızlı bir gelişme ve şiddetli bir rekabet yaşanmıştır. potansiyel müşteriler. Ayrıca, müşteri memnuniyeti, özellikle firmalar arasındaki rekabetin çok yoğun ve azalmadığı bankacılık iş ortamında, müşterileri çekmek ve elde tutmak için çok önemlidir. Şikayetlerin ele alınması, bekleme süresi ve müşteri ilişkileri yönetimi gibi müşteri hizmetlerine azami özen gösterilmelidir.

Anahtar kelimeler: marka bilinirliği, mağaza imajı, hizmet kalitesi marka değeri, Sierra Leone

Table of Contents

Approval	2
Declaration	3
Acknowledgments	4
Abstract	5
ÖZ	6
Table of Contents	7
LIST OF FIGURES	10
LIST OF TABLES	10
CHAPTER I	13
Introduction	13
Background to the Research	13
Background to Sierra Leone Banking Industry	14
Statement of the Problem	16
Research Purpose, Specific Objectives and Research Questions	17
Scope and delimitation of the Study	
Significance of the Study	
Organization of the Research	20
CHAPTER II	21

Literature Review
Introduction
Theoretical Review
Associative Network Memory Theory21
Flow Theory
Brand Awareness
Store Image
Service Quality
Brand Equity
Review of current Literature relevant to the research questions
Research Gap34
Hypotheses Development
The relationship between Brand Awareness and Brand Equity
The Relationship Between Store Image and Brand Equity
The Relationship Between Service Quality and Brand Equity
Conceptual Framework
CHAPTER III
Research Methodology
Research Design
Sampling and Sample40

Data Collection Procedures41
Research Materials
Data Analysis Procedures44
Ethical Considerations45
CHAPTER IV46
Research Results
Descriptive Statistics
Brand Awareness
Store Image
Service Quality
Brand Equity
Demographic Characteristics of Respondents
Gender
Age51
Educational Level
Marital Status
Position held in the organization53
Kindly select your organization55
Correlation Analysis
Exploratory Factor Analysis

Hypotheses Testing	58
The Relationship between Brand Awareness and Brand Equity	58
The Relationship between Store Image and Brand Equity	60
The Relationship between Service Quality and Brand Equity	61
Overview of Hypotheses Testing	62
CHAPTER V	64
Discussion and Conclusion	64
Discussion	64
Conclusion	65
Recommendation	67
Reference	68
Appendix X	81
QUESTIONNAIRE	81
Turnitin Similarity Report	88
ETHICAL COMMITTEE APPROVAL	89

LIST OF FIGURES

Figure 1 Conceptual Framework

LIST OF TABLES

Table 1 population, sample size and sampling technique of respondents amongst selected
Banks in Sierra Leone41

Table 2 The Cronbach's Alpha for Research Variables	43
Table 3 Correlation Coefficient Scale	44
Table 4 The Distribution of The Questionnaire on The Research Sample	46
Table 5 The Degree of Approval of the Questionnaire Paragraphs	47
Table 6 The Mean Scores for the BA Items	48
Table 7 The Mean Scores for the SI Items	48
Table 8 The Mean Scores for the SQ Items	49
Table 9 The Mean Scores for the BE Items	50
Table 10 Sample Distribution by Gender	51
Table 11 Sample Distribution by Age	51
Table 12 Sample Distribution by Educational Level	52
Table 13 Sample Distribution by Marital Status	53
Table 14 Sample Distribution by Position held in the organization	53
Table 15 Sample Distribution by Duration spent with the organization	55
Table 16 Sample Distribution by Kindly select your organization	56
Table 17 Correlation between Variables	56
Table 18 KMO and Bartlett's Test	57
Table 19 Regression analysis of BA on BE	59
Table 20 Regression analysis of SI on SE.	60
Table 21 Mediation analysis of SQ and BE	61

Table 22	Overview	of Hypotheses	Testing62	2
			0	

CHAPTER I

Introduction

This chapter will be presenting the background to the study on the impact of brand awareness, store image, and service quality in promoting brand equity in the banking industry of Sierra Leone. It will cover the background of the case study (the baking industry in Sierra Leone and most especially the four (4) selected banks for this study) of this research. The chapter identifies related questions and objectives of brand awareness, store image, and service quality in promoting brand equity and the significant of the study. It will present the statement of the problem that derive to the objectives and research questions.

Background to the Research

Rivalry and the mission to separate associations from each other has driven a few associations to burn through billions of dollars on branding (Maddox, 2009). The gigantic assets firms are focusing on branding embody the developing obligation to branding exercises. Reports in the U.K. show that however much £2million is spent yearly on branding and its connected exercises by certain organizations (Vrontis and Thrassou, 2006). Further reports affirm that in many nations, monetary assistance firms spend a ton on branding. Papasolomou and Vrontis (2006) uncover that in the U.K. for example, almost £1 billion is spent yearly on brand building exercises by monetary assistance firms. This expanding speculation on branding is not astounding given the apparent capacities a brand can offer on associations that form them (Scott and Bennett, 2014). In addition, the mechanical refinement of the present proficient shopper (Zhang et al., 2010), combined with a noticeably packed serious market climate (Chellapa, Sambamurthy and Saraf, 2010), make it basic for banks to separate themselves from their opposition through branding. Thusly, it comes as little amazement that branding has become a significant thought in financial today (Papasolomou and Vrontis, 2006).

In Africa, the Sierra Leone financial industry is without a doubt quite possibly the most serious enterprises in the nation (PWC, 2014). The area as of late has encountered a heap of marking exercises, which incorporate the section of commercial banks, the separate

ingestion of indigenous banks. Moreover, eight bank brands were addressed in premierbrand.com's best 50 important brands in Sierra Leone in 2014. This vouches for how far the financial business has come since the monetary area changes of the last part of the 1990's which was portrayed by the section of unfamiliar banks, remarkably banks from Sierra Leone's serious neighbor.

In Sierra Leone, as the conduits opened and new banks overwhelmed the Sierra Leone banking industry, rivalry expanded, working up an expanded mindfulness on the requirement for banks to accept branding as methods for separation and endurance. Moreover, the presence of extreme rivalry makes client steadfastness a result that most banks want. Banks try to create brand steadfastness as methods for battling client stir. Various investigations have made it clear the significance of keeping and keeping up faithful clients. The agreement is that it costs firms more to get new clients than to hold existing clients. Client unwaveringness has along these lines become a significant thought for banks who have increased determination to keep hold of clients. This is not astounding, given that unwaveringness is vital for any reason and is a vital supporter of the development of each association.

BE is likely one of the most important marketing concepts in both business practice and academic research (Atilgan, Aksoy & Akinci, 2007). This is because successful brands can allow companies to have a competitive advantage (Ms. Michele et al., 2016). Given a much broader array of product choices offered in the current market and ubiquitous marketing efforts, consumers often turn to a favorite brand to facilitate their purchase decisions. A brand includes a name, symbol, design, or experience that helps consumers identify products, services, or differentiate offerings among competitors. In this vein, this work will analyze impact of brand awareness (BA), store image (SI), and service quality (SQ) in promoting brand equity (BE) in the banking industry of Sierra Leone.

Background to Sierra Leone Banking Industry

The financial sector in Sierra Leone is one of the shallowest in the region, and has been declining. Credit to the private sector declined from 7.5 percent of GDP in 2011 to 5.8 in 2016 and to 5.1 in 2017. The major driver of this has been increased lending from the

banking sector to the central government, which increased from 5.5 percent in 2011 to 16.18 in 2016 and 17.6 percent in 2017(The World Bank, 2018).

The banking sector is dominated by commercial banks. The banking sector in Sierra Leone includes 14 commercial banks, 17 community banks 59 financial services associations (FSA), 13 MFIs (2 deposit taking and 11 credits only), and three mobile money operators. The commercial banking sector accounts for 99 percent of all financial sector assets. There are 11 foreign banks, two state owned banks, and one domestic banks. Total assets of the commercial banking sector are Le5.29trillion, and account for over 99% of the assets in the financial system. The 11 foreign banks account for most than 60% of the industry's total assets (The World Bank, 2018).

Outside of the two state-owned banks, the banking system is broadly stable. The Capital Adequacy Ratio (CAR) for the banking system increased from 30.7 percent to 34.1 percent in 2017, indicating that the system has adequate buffers in the case of losses. Asset quality has also improved in recent years, although the absolute level remains problematic. The level of non-performing loans (NPLs) peaked at 33.4 percent at end2014 and declined to 23 percent at end-2016 and 18 percent at end-2017. Profitability has also improved, with return on assets (ROA) increasing from 2.9 percent at end-2016 to 5.3 percent at end-2017 and return on equity (ROE) increasing from 22.3 percent at end-2016 to 25.6 percent at end-2017. Commercial banks are also growing, with new deposit accounts growing by on average in recent years partially due to an increasing branch network (The World Bank, 2018).

Two large state-owned banks pose challenges to financial stability. Despite the entrance of a number of foreign banks over the past decade, the two state-owned banks (Rokel Community Bank - RCB and Sierra Leone Commercial Bank - SLCB) continue to be key players with 28.6 percent of assets, 36.2 percent of deposits, and 23.8 percent of credit. Both RCB and SLCB had to recognize large and long-standing asset problems in 2014, which resulted in an erosion of their capital base and measures by the authorities to initiate their restructuring. This was the major driver of the system-wide NPL increase mentioned previously. BSL has intervened in the banks and has temporarily taken over management of the two institutions and has put in place mechanisms to limit new corporate lending.

However, the underlying questions about their business model and their viability remains and needs to be addressed. The government is currently determining how to proceed with the restructuring of these two banks. The outcome of this restructuring will have significant impact on the financial stability of the country, which amongst other things, will likely increase risk taking amongst banks and decrease the cost of finance overall for enterprises (The World Bank, 2018).

The 17 community banks, 59 FSAs, 2 deposits taking and 12 credits only MFIs are the primary branch based financial service providers in rural areas. Community banks, FSAs, and MFIs primarily cater to micro and small entrepreneurs and enterprises. These institutions also have a significant focus on agricultural lending and on rural areas. Women also make up a significant portion of loan recipients, 44.05 percent in the case of FSAs. The 17 community banks and 59 FSAs account for less than 1% of total assets in the financial system and have had uneven performance in recent years. BSL has indicated that this was due to poor corporate governance, internal controls, lack of basic infrastructures, weak information and communication technology. The BSL has increased on-site and offsite inspections of these banks and the International Fund for Agricultural Development (IFAD) has supported the restructuring of six (6) Bank of Sierra Leone pioneered community banks and also established eleven (11) more community banks and fifty-one (51) Financial Services Association (FSAs). The two deposit taking MFIs and 12 credits only MFIs. The two-deposit taking and twelve credits only MFIs have less than 1% of total financial sector assets and are relatively small, but have a total of 101 branches and a wide geographic reach. BSL took over supervision of these institutions in December 2015 (The World Bank, 2018).

Statement of the Problem

Due to the fast change in the global market and increase competition, the management of the brand has become of crucial importance. The building of strong brand equity is the topmost priority of many banks, but attaining this objective is not always an easy task because the products and services of many banks are similar and their means of distributions are alike. Brand awareness, store image, service quality, and brand equity are the only possible means by which customers can differentiate one brand from another. According to Denmark (Denmark, 2010) the concept of the corporate brand is vital and draws synergy between organizational resources (human, fixed resources, tangibles, and intangibles) and the strategic objectives of the organization to achieve success among competitors. In this regard, after achieving success through corporate branding, maintaining and managing the brand's reputation becomes integral to be the market leader (Denmark, 2010).

The scenario playing out in the Sierra Leonean banking industry is that of a battle for supremacy. With an influx of multinational banks in Sierra Leone, the level of competition in the banking industry has reached an all-time high. Building strong brands has therefore become the imperative for banks and other organizations in Sierra Leone. Though a consensus exists on the relevance of branding, it appears banks still yearn for more knowledge about how their branding efforts can be channel towards specific outcomes such as customer loyalty. Dobni (2002) noted that all banks have one goal in common - to outperform rivals and stay on top of the competition. This observation cannot be far from the truth as far as the Sierra Leone banking industry is concerned.

Therefore, the author has chosen to carry out research to indicate the importance of these three dimensions of BE (BA, SI, SQ) in the banking industry of Sierra Leone.

Research Purpose, Specific Objectives and Research Questions

Generally, the purpose of the study is to examine the impact of brand awareness, store image, and service quality in promoting brand equity in the banking industry of Sierra Leone.

The study will be guided by the following specific objectives:

- To examine the relationship between brand awareness and brand equity in the banking industry of Sierra Leone.
- To ascertain the relationship between store image and brand equity in the banking industry of Sierra Leone.
- To evaluate the relationship between service quality and brand equity in the banking industry of Sierra Leone.

• To study the factor structure analysis of brand awareness, store image, service quality, and brand equity in the banking industry of Sierra Leone.

The study will be guided by the following research questions:

- 1. What is the relationship between brand awareness and brand equity in the banking industry of Sierra Leone?
- 2. What is the relationship between store image and brand equity in the banking industry of Sierra Leone?
- 3. What is the relationship between service quality and brand equity in the banking industry of Sierra Leone?
- 4. What is the relationship between the factor structure analysis of brand awareness, store image, service quality, and brand equity in the banking industry of Sierra Leone?

Scope and delimitation of the Study

The subject scope of this study intends to examine the impact of brand awareness, store image, and service quality in promoting brand equity in the banking industry of Sierra Leone. The study will be carried out at in Sierra Leone focusing on the banking sector. I will therefore, use at least four banks in Sierra Leone since they play a vital role in brand awareness, store image, and service quality in promoting the brand equity process in the country. The researcher will involve 100% staff within Freetown. The study will capture brand equity and it dimensions from the period between 2015 to 2019.

Significance of the Study

The outcome of this study is expected to contribute significantly towards the banking companies on the effects of brand awareness, store image, service quality, and brand equity. In the banking sector in Sierra Leone, it has been observed that the leading banks are "brand conscious", displaying an awareness of the importance of brand building and the benefits it offers to them. Many banks in Sierra Leone appear to have invested huge sums of money in the visual as well as the intangible dimensions of branding and their presence is heavily felt in the industry, especially as the mention of their names evoke

certain expectations from customers in the industry. Branding indeed offers many benefits to banks, but it is widely believed that the bottom line for banks when it comes to branding is to create an outcome of loyalty that would reduce rates of customer defection and increase rates of retention and loyalty. This study is therefore important as it offers a unique perspective on how specifically branding leads to customer loyalty in the banking industry in Sierra Leone.

The findings of this study are relevant for scholars as well as industry practitioners. Below are some factors that highlight the significance of this study:

The study contributes to theory by providing an insight into how banks can use branding as a means to achieve specific outcomes such as loyalty. The use of structural equation modelling in the analysis of the data ensured that the best model that fits the data was chosen, hence guaranteeing that the results obtained in this study are credible and reliable. This dissertation thus contributes to academia by closing the knowledge and context gap on branding and customer loyalty from a developing country perspective and in a West Africa setting. This study also serves as a reference point for future research on the research topic or similar research topic. Furthermore, this thesis can be used as a source document that can engender debate and stimulate further discussions on the issue of bank branding and customer loyalty.

In the wake of increased competition and technological sophistication and advancement, this study is a welcome relief to industry practitioners and marketers in the banking industry in Sierra Leone as it provides information on specific branding measures that can be implemented to reduce customer churn and increase customer retention and loyalty rates. Findings of this study are also relevant to advertising agencies as it informs them on how to develop content for adverts for banks and their customers. Particularly revelatory was the fact that creative adverts and humor adverts played key roles in banks' marketing programs that inevitably contribute to customer loyalty.

Organization of the Research

The study comprises of five chapters. The content of each chapter is outlined and concisely described below:

Chapter one introduces the study by illuminating the research context, outlining the research objectives and research questions. Furthermore, the scope and delimitations as well as the contributions of the study are contained in this chapter.

Chapter two presents a review of the relevant literature in branding equity and it three dimensions. Furthermore, the conceptual framework for the study is presented in this chapter, detailing the various hypothesized relationships and highlighting the research gap.

Chapter three provides a detailed description of the methodology used in achieving the study's objectives and research questions. It also provides a profile of the respondents to be selected, and the validation for the methodological choices including the data analysis techniques.

Chapter four focuses on the analyses and discussion of the empirical results. It also presents the major findings of the study, discussing the key results of the study in relation to the literature.

Chapter five outlines the major conclusions of the study. It further outlines the theoretical contributions to academic knowledge, provides a number of managerial implications, and concludes with recommendations for further research.

CHAPTER II

Literature Review

Introduction

This chapter reviews relevant literature on contemporary issues and dynamics of brand awareness, store image, and service quality and explores relevant theories that promote brand equity in the banking industry of Sierra Leone. The chapter also identifies existing literature gaps; develops relevant hypotheses and builds new theories and provides a conclusion to the chapter.

Theoretical Review

The importance of good theory in research is stressed. The aim of this paper is to look at how theory has affected brand equity analysis. It aims to respond to the study questions in particular. The researcher used a systematic analysis of papers written in the journal of brand equity to address the research questions. This was done to map and measure the current academic landscape. This research will be based on seven hypotheses that are applicable in the sense of brand recognition, store image, and service efficiency, and will examine relevant theories in fostering brand equity.

Associative Network Memory Theory

The associative network memory theory (ANMT) is based on how memory works (Srull & Wyer, 1989). Indeed, the theory assumes that memory is made up of nodes, which are small bits of data (Smith, 2004). Via external information gathered from external sources or internal information collected from long-term memory, a node acts as an activator for other nodes (Keller, 1993). When a person receives information from internal or external sources, it is decoded into an abstract mental language and assigned to a network knowledge structure node (Kreuzbauer & Malter, 2005). Memory values, according to Keller (1993), influence brand awareness and knowledge, and therefore brand equity.

Understanding the structure of brand awareness is important because it affects how consumers think about a particular brand. ANMT, on the other hand, proposes that semantic memory is made up of a set of nodes linked together by associations (Keller, 1993).

Keller (1993) pointed out that brand associations can be studied with the help of an associative network memory model. Consumer associative network mostly takes the side of consumer memory and information stored in memory (Henderson et al., 1998). The most widely accepted conceptualizations of memory structure involve some types of associative model formulation (Anderson, 1983; and Srull and Wyer, 1989). John et al. (2006) found that strong, favorable and unique associations can produce brand association networks; and if we add these associations, it will support in forming the brand association map. Brodie and Benson (2016) pointed out that the branding process in connection with the country of origin or any country linked association is a network model extension. Jevons (2005) mentioned that brands are more than brand management, and in many contexts, we can question whether any initiative can be called a brand.

Associative network memory model views semantic memory or knowledge as consisting of a set of nodes and links. Nodes are warehoused information connected by links that vary in strength. A 'spreading activation' process from node to node determines the extent of retrieval in memory (Collins and Loftus, 1975; Raaijmakers and Shiffrin, 1981; and Ratcliff and McKoon, 1988). A node becomes a potential source of activation for other nodes, either when external information is being encoded or when internal information is retrieved from long-term memory. Activation can spread from this node to other linked nodes in memory. When the activation of another node exceeds a threshold level, the information contained in that node is recalled. Thus, the strength of association between the activated node and all linked nodes determines the extent of this 'spreading activation' and the particular information that can be retrieved from memory.

Flow Theory

Flow theory received a growing attention in the last years as a basis to understand the facilitators for develop compelling experiences in online environments. The origin of this

theory can be traced back in the papers of Csikszentmihalyi. While interviewing certain professional and amateur dancers, chess players, rock climbers, surgeons, and many others that expressed a deep devotion for their preferred sort of activity, Csikszentmihalyi selected a special feeling, called "optimal experience". During the interview sessions, Csikszentmihalyi found that individuals verbalize it as "being in the midst of a flow". Csikszentmihalyi (1975; 1990) called this holistic experience flow.

Flow theory is one of the two theories of intrinsic motivation developed in positive psychology paradigm (Seligman and Csikszentmihalyi, 2000). The other theory (Self-Determination Theory) was developed and presented by Deci and Ryan (1985). In fact, the theory of optimal experience or flow is not only motivational. Csikszentmihalyi and his colleagues were engaged in serious efforts to present flow experience as: (1) an cognitive artifact (the starting point is the center of attention), applied for the description of holistic personal development (Csikszentmihalyi, 1978, 1975), (2) an important factor in the evolution of bio-culture and selection (Csikszentmihalyi, 1990; Massimini and Delle Fave, 2000), (3) a theory of creativity, work well done and talented adolescents (Csikszentmihalyi, 1996; Csikszentmihalyi, Rathunde and Whalen, 1993; Gardner, Csikszentmihalyi and Damon, 2001), (4) a psychological theory of development (Csikszentmihalyi, 1990; Csikszentmihalyi and Larson, 1984); (5) a basis for psychological rehabilitation practice (Delle Fave and Massimini, 2004, 2005).

Despite the variety of interpretations, in this study the flow experience is discussed as a motivational paradigm. Flow theory was first applied to online environments by Hoffman and Novak (1996). They extended the universal applicability of flow to computer-mediated environments and suggested that "optimal experience" can contribute to the success of online marketers by creating exciting experiences to consumers. Because the Internet is an interactive environment, it provides opportunities to customers to experience flow during their navigation (Chante al., 1999). Because of its marketing outcomes, flow experience is a desirable consequence of online interaction between the consumer and firms.

Flow is a cognitive and affective condition that occurs when people are engaged in the task at hand: it is described by complete focus, complete participation, lack of selfconsciousness, and optimal enjoyment (Csikszentmihalyi, 2000). The key causal factor for experiencing flow, according to flow theory, is that the perceived challenges of the task demands and the skills one possesses to deal with these challenges are both relatively strong and in balance. Flow is typically characterized by the following aspects: (a) clear goals; (b) immediate and unambiguous feedback (Lambert & Csikszentmihalyi, 2019); (c) balanced skills versus challenges; (d) sense of control over the task at hand; (e) high but subjectively effortless attention; (f) distorted sense of time perception (time moves faster or slower than usual); (g) a merging of awareness and action takes place; (h) a loss of self-awareness; and (i) an 'autotelic' motivation, namely, engaging in the task itself is perceived as rewarding. Since high-challenge, high-skill situations are more common in work and formal leisure activities in everyday life, flow is considered to be more regular and intense in these settings (Shernoff et al, 2017). This is significant because telecommunications is an essential service provider in people's daily lives, providing both work and leisure functions as well as comfort and pleasant experiences. As a result, research into the impact of flow experience, as provided by various telecommunications, on customer preference will help us better understand brand identity and purchasing intention promotion. This approach can extend the investigation of flow theory from the hedonic-utilitarian and online shopping perspective in past studies (Lim, 2014) to the branding perspective and telecom in the present study.

Brand Awareness

According to Romaniuk, Wight, and Faulkner (2017), brand awareness/recognition is a customer's ability to identify or remember that a certain brand belongs to a specific product category. Brand awareness is described as the ability of customers to recognize (identify) a brand based on its existence and traces in their minds (O'Guinn, Allen, & Semenik, 2009). From the above meanings, it can be inferred that brand awareness refers to being conscious of the presence of a brand and its connection to a specific product. According to Kahneman (2012), current research shows that even one presence of a brand name in a consumer's environment leads the customer to think, "I know this brand," when the

customer sees it again. It's also worth noting that many promotional campaigns are built on this premise, with messages aimed at cementing the brand name in the minds of current and future customers. As a result, brand recognition measurement is becoming increasingly important in modern business practices.

According to Torelli (2013), there are three forms of brand recognition measurements:

- a) spontaneous awareness, which shows the percentage of customers who can name a given brand without the help of an inter-viewer; this metric determines the real field of customers' choices at the same time
- **b**) Top of mind awareness, which indicates the percentage of respondents who name a specific brand as the first in the study (the first that comes to mind); this indicates that the brand is well-known among buyers.
- c) Supported awareness, which indicates a poor relationship with a given brand by indicating what percentage of respondents declares brand information after it is listed by the interviewer.

In practice, some scholars like Farris et al, 2010 pointed out that brand awareness reflects the success of a company's marketing efforts. It is important to note that shaping brand awareness entails cementing the brand's presence in the minds of consumers, which translates into decisions about the production of the offer, the selection of specific activities, and the cost analysis of marketing activities that should be carried out.

Brand awareness is needed for the development of a strong brand (Buil, de Chernatony, & Martnez, 2013). Brand recognition sets the brand apart from the competition and contributes to brand selection (Valavi, 2014). Furthermore, according to Balaji (2011), brand recognition affects the strength of brand relationships in the minds of customers. Most consumer behavior models argue that brand recognition is the first and most important step in selecting a brand. Specifically, brand recognition is regarded as a critical component of brand equity (Azad, Al Muzahid & Kamal, 2013).

Furthermore, brand awareness is critical in establishing a brand in the minds of consumers because customers make buying decisions based on their expertise, awareness, or experience with a particular brand. As a result, consumers are more likely to buy again because they are confident in the product's efficiency. As a result, brand recognition is linked to brand loyalty when it comes to making choices based on consumers' thoughts. Customers' brand loyalty grows stronger and firmer as they become more knowledgeable of the brand.

Brand awareness refers to the level of consumer recognition, acceptance, and recall of a brand in any case (Percy and Rossiter, 1992; Perreault et al., 2013). According to Aaker (1991) brand awareness is "the ability of a potential buyer to recognize or recall that a brand is a member of a certain product category". Keller (2009) has stated that brand awareness is about track or crowd power in consumers' memories that reflect consumers' ability to remember or recognize a brand in different conditions. Brand awareness reduces the time and risk that consumers will spend searching for the product that they will buy (Verbeke et al., 2005). In this respect, consumers are expected to choose the brand that they have information. Aaker (1996) has stated that brand awareness consists of four levels: brand recognition, brand recall, top of the mind brand and dominant brand. Brand recognition is associated with brand familiarity of the consumer while brand recall is the thinking the brand at first when a range of product is introduced (Farjam and Hongyi, 2015). Being a brand that comes to mind at first refers to becoming the most aware of brand in product category. The level of brand dominance refers to the level at which the brand replaces the product category (Aaker, 1996).

Store Image

Brand image is described by Zhang (2015) as "a system of images and thoughts occurring in human consciousness, expressing information about a given brand and basic attitude toward it." From this concept, it can be deduced that brand image is formed by a collection of extraordinary features that are unique to a given brand and trigger differentiation while also ensuring market awareness. At the same time, Pars & Gulsel (2011) emphasize that brand image is an impression formed as a result of a variety of factors (e.g., associations with a given brand name, purchasing experience, company reputation, forms and measures of advertising, promotion, and so on), implying that it is a complex, heterogeneous, and abstract category from the perspective of various recipients.

The reason for their study is that store brands' image positioning is dependent on that of the store, implying that retailer or store image, and thus store credibility and commercial value, influence store Brand Equity to a large extent (Beristain & Zorrilla, 2011). The explanation for this may be that when the manufacturer's information is not given, the retailer's or store's image plays an important role as a consumer guarantee. Customers will be able to rely on the store's image in this way, lowering the perceived risk of buying store brands. Given that the extrinsic cue generated by the store image should be more appropriate for store brands than for producer brands, this proposed model will enable us to identify the effect of store image on store Brand Equity and, as a result, on store brands' purchasing intent. According to Ailawadi and Keller (2004), the image of the retailer or store in the minds of customers is the foundation of the store Brand Equity; therefore, retailer and store image influences customer preferences and buying behavior, and thus, retailer and store image becomes a significant foundation for their store Brand Equity. Therefore, I propose considering these store image variables in the customer-based store Brand Equity model, assuming that store reputation and store commercial image, would influence store Brand Equity.

Store image is evaluated based on the overall attitude or impression of the consumer obtained through the internal and external characteristics of the store (Wu et al., 2010). The characteristics being evaluated by the consumers are service quality, store atmosphere, product variety, and product prices (Ailawadi and Keller, 2004). When consumers have a positive evaluation of the store image, then the evaluation has a positive impact on private label products (Wu et al., 2010). Similarly, the findings of Liljander et al. (2009) say that the store image has an indirect impact on purchase intention through perceived risk and perceived quality. This is confirmed by Bao et al. (2011) that perceive store image has a positive impact on consumers' purchase intention. Schiffman and Kanuk (2007) state that consumers will choose products according to the expected brand image, so that consumer attitudes towards brand image have a positive influence on purchase intention (Chao, 2015). In general, a better store image will lead to higher consumer purchase intention (Cretu and Brodie 2007; Keller 1993) in Chao (2015). Furthermore, brands play an important role in the purchase decision process, because the higher the brand image will reflect on the level of product quality and increase purchase intention.

Likewise, a positive store image will reduce consumer perceptions of purchasing risk (Batra and Homer, 2004) in Chao (2015).

Service Quality

Service quality is sometimes interpreted as a measure of how well the standard of provided services meets consumer expectations (Santos, 2003). Service quality (SQ) is elusive and abstract, unlike products quality, which can be calculated with some objectivity. Since objective metrics are lacking, businesses must rely on customers' expectations of service quality to identify their strengths and/or shortcomings and implement effective strategies.

This necessitates the development of managerially useful and psychometrically sound scales to assess service quality. Service quality is critical to a company's competitive advantage. The assessment of service quality has been extensively researched in order to assist practitioners in effectively managing the delivery of high-quality service. Service quality is assessed to assess service efficiency, identify service issues, manage service delivery, and provide a foundation for corporate and employee incentives.

Gefan (2002) defines the quality of services as a subjective comparison that customers make between the quality of services that they prefer to receive and what they actually achieve. The literature relating to the definition of the quality of services is widely classified into two technical and functional perspectives. The technical perspective gives the nature of the provided services, while the functional one investigates the mode of services (Gro noos, 1983). In General, marketers defend the functional approach when they want to measure the quality of a company's services. The main reason for this preference is based on the argument that a consumer has typically a different perception of the quality of services compared to a technical expert. Therefore, for measuring the consumer's perception of the quality of services, the use of technical method is not appropriate (Donabedian, 1980). Some researchers have defined the quality of services based on different theoretical hypotheses. For example, Parasuraman (1990) believes that the quality of services consists in the existing difference between the expectations and the customers' perception of received services. In recent three decades, the different models were provided by the experts and scientists for measuring the quality of services. One of the most prestigious and the most famous of them is the measurement model of quality of services of Parasuraman, Zeithaml and Berry (1991) called the SERVQUAL model. SERVQUAL is regarded as an acceptable model for the measurement of an organization's quality of the services. As a tool for measuring the quality of services, Parasuraman et al. (1991) have provided a questionnaire. The SERVQUAL questionnaire includes 22 items that measure five dimensions of the quality of services. These dimensions consist in the tangible factors, reliability, responsiveness, assurance and empathy (Parasuraman et al., 1991).

- **Tangible factors**: tangible factors include appearance, physical possibilities, equipment, employees and communicative commodity. All these dimensions provide an image that the customer considers for evaluating the quality; like the amenities in the physical environment of the organization.
- **Reliability:** reliability means the ability to perform services in a safe and reliable way, so that the customer's expectations are satisfied. For example, customers who go to a restaurant have the expectations such as doing quickly the orders and cleanliness of the surroundings. Other meaning of the reliability is keeping commitments. In other words, if the organization promises some services in the field of time of presenting services, way of doing those and cost of them, it has to keep them.
- **Responsiveness:** responsiveness refers to the desire to assist customers and provide quickly the services. This dimension of the quality of services emphasizes on demonstrating sensitivity and vigilance in exchange for customer requests, questions and complaints. For example, in a restaurant if a customer gets into difficulty in connection with the food order request and requests for help, the length of time that he/she should wait for the fixing it, will represent the amount of employees' responsiveness.
- Assurance: the reliability of the knowledge and employees' polite behavior refers to competence and merit of services to suggest assurance and self-assurance. This dimension of the service quality is especially important in those cases in which the risk is high. For example, according to the sensitivity of the lives of people in food

intake, in the restaurant and fast-food industry this component is of great importance.

• **Empathy:** empathy means the individual care and attention to the services that are provided to their customers. It means that despite the individuals' spiritually each of them is conducted in a special manner, so that customers are convinced that the organization has understood them and they are important for the organization.

Brand Equity

Perceiving a brand in terms of its worth, or equity, is critical in the process of developing a competitive advantage for a company and determining its market place in relation to other brands. According to a review of the literature on the subject, there are several approaches to defining brand equity. As a theoretical basis for their considerations, they typically presume two separate points of reference. The first is concerned with the financial aspects of brand equity, while the second is concerned with improving and increasing the effectiveness and efficiency of marketing activities, especially in the communication sector.

In today's business climate, establishing good brand value is critical. Strong distribution networks, product brand extensions, and price elasticity may all be aided by consumerbased brand equity (Kay, 2006). As a result, marketers are paying close attention to brand future prospects and are constantly looking for effective ways to build good brand value in order to stay competitive or avoid becoming redundant (Wilcox et al, 2008). To address these issues, marketers and analysts have emphasized the importance of brand equity (Faircloth, Capella, & Alford, 2001), since customers are typically loyal to only a few brands (Thomson, MacInnis & Park, 2005). Efficient branding that results in positive customer evaluations is, without a doubt, a critical component of brand success. Keller's consumer-based brand equity model, on the other hand, offers a structure for how consumers' responses contribute to brand equity growth (Kakati & Choudhury, 2013).

According to Aaker (1991), brand is: a set of assets and obligations which is linked with the brand name and symbol; it is added to or lowered from the value that is provided by a product or service to the company or the company's customers. Keller (1993) suggested that consumers' brand awareness is based on the set of dependency related to properties, advantages and general characteristic. In the other words, the brand equity is defined as the customers' subjective and intangible evaluation of brand (Ryan et al., 2005). The brand equity is a value that is created by increasing the effectiveness of marketing activities, creating a higher degree of brand preference, the desired purchase intent and increase in the market share of a firm, (Bailey & Ball, 2006; Aaker, 1996; Berry, 2000; Chang & Chiang, 2006; Farquhar, 1996). Keller (1993) sees the consumer-based brand equity as the recognition of the brand impact relating to the consumer on the consumer's reaction towards the company's compound marketing strategy. The four dimensions of the brand equity that was provided by the Aaker (1991) are as follows: the perceived quality, brand awareness, brand associations, brand loyalty.

- **Perceived quality:** Zeithaml (1998) has defined the perceived quality as customers' judgment towards the advantage, superiority, credit and the difference of a brand compared to other rival brands. Perceived quality affects not only on other dimensions of brand, but also the classification of products from a customer's perspective (Aaker & and the, 2000).
- **Brand associations**: according to Aaker (1991), brand associations consist in anything in memory that is related to brand. Consistent and brand association may be observed as the forms and characteristics related to a product or to the independent aspects of a commodity itself that the stays in customer's memory (Chen, 2001).
- **Brand awareness**: Aaker (1991) defines the brand awareness as an ability of a potential buyer for recognizing or reminding a brand that is a member of the particular commodity class. In fact, awareness is defined an area in which the customers recognize and remind the brand and can identify the products and services related to the brand (Aaker & and the, 2000).
- **Brand loyalty**: according to Aaker (1991), brand loyalty is a situation that indicates how is likely a consumer to change a brand, especially when that brand causes a change in the price or features of products and services. Loyalty is defined as a behavioral response as well as a physiological function. Brand loyalty is a

descriptive variable that reflects the unique differences of behaviors and purchase style of customer in a specific product class (Gounaris & Stathakopoulos, 2004).

Review of current Literature relevant to the research questions

According to Sulieman (2013), the quality of banking services offered to customers has improved, and it can be used as a foundation for achieving service excellence in order to compete, where distinguished service has become the basis for differentiation between banks. The definition of service quality is to match and adapt to the needs, which means that service organizations, such as banks, must have unique criteria and specifications for the services they offer to customers. Service quality, on the other hand, is described by Al-Mahiaoi (2006) as "the criterion for a match between the actual performance of service and customer expectations for this service". With the changing market climate, banks are attempting to go beyond efficiency by offering distinguishing banking services, thus providing a sense of satisfaction to customers with the service acquired and meeting their expectations.

According to Moala (1998), cited in Hirut (2015), a study was conducted to assess the quality of banking services provided by Jordanian commercial banks from the perspective of their customers, and the study discovered that the actual output of banking services is poor when compared to the level of quality that customers expect from their banking services. The study recommended that the departments of commercial banks should develop programs to enhance the quality of their services.

According to Zarour (2003), cited in Hirut (2015), the study found a relationship between industrial company brand equity and some dimensions of quality banking service and the prices of these services. Banking departments should improve the quality of their services and prices, according to the report.

According to Abu Muammar (2005), who conducted a study titled "banking service quality measurement in banks operating in the Gaza Strip, Palestine," the aim of the study was to determine the relationship between service quality and customer-based brand equity for customer satisfaction. The study discovered that in order to provide the best banking services and achieve brand equity and satisfaction, it was necessary to understand customer expectations. Furthermore, customer loyalty and perceived quality are heavily reliant on the quality of banking services being investigated. The study recommended that the customer-bank relationship be strengthened and that banks be more interested in speed, accuracy, and the provision of banking services.

Study on "assessing the efficacy of the quality of health services in the State of Kuwait's Jahra hospital," according to Al-Saidi (2007), cited in Hirut (2015). The study discovered that the sample study's response to the efficacy of quality services was strong, with statistical differences based on educational level. This research suggested that a specialist quality control center be established in the hospital in order to gain customer-based brand value for the center.

Abu Zaid (2009) conducted a study titled "effect of service quality on customer loyalty and satisfaction in commercial banks." The aim of this study was to compare the degree of consumer-based brand equity to the reality of quality service in commercial banks in the West Bank. The results showed that customer satisfaction, loyalty, and perceived quality had increased. According to the report, banks should continue to hold staff training sessions, poll customers about services offered, and have access to Arabic and foreign banking experience.

Addams, (2000) Study entitled "Service at Banks and Credit Unions," the study found that credit Institutions are better than banks on providing quality banking service as the Bank does not identify customer needs, the study recommended the need to pay attention to the quality of banking services, and to identify the needs of customers. Hunjra et al., (2011) Study entitled "the relationship between customer satisfaction, and quality of service in Islamic banking in Pakistan". This study aimed to highlight the relationship between customer satisfaction and quality of service for Islamic banks. The study found a relationship between customer satisfaction and building the quality of bank service, the study also ensures that the effect of the response to the needs of the customers and satisfaction are of the most influential variables in Pakistan.

Research Gap

In this study, the relationship between Brand Awareness, Store Image and Service Quality is established basing on how they impact on Brand Equity. This is based on international literature. From the above explained literatures, it is evident that a good number of researches on the selected variables capacity building have been done extensively oversees. However, there is evidence which shows less has been done here in Sierra Leone specifically to the banking industry. Therefore, this study will be conducted in Sierra Leone, Freetown. Hence, the findings from this study, recommendations and conclusion covered this gap. The external dimension of brand equity (roles or functions, model and impact) that is critical in Brand Awareness, Store Image and Service Quality is vitally important. To date, no linkage has been established between Brand Awareness, Store Image, Service Quality and Brand Equity and this study will help to bridge this gap.

Hypotheses Development

The relationship between Brand Awareness and Brand Equity

Brand equity measures are generally classified into three subsets: customer mindset measures, brand performance measures, and shareholder value measures (Keller and Lehman 2003). Customer mindset measures gauge customers' general attitude toward a brand and include two important components: brand awareness and brand association. Brand association refers to any brand knowledge relating to the brand in the customer's mind. This knowledge represents overall brand equity in the customer's mind. The following discussion considers customer mindset brand equity as synonymous to brand association. The second group of brand equity measures, called product market performance measures, assesses the brand market performance resulting from customer-mindset measures and include dollar sales, volume sales, revenue premium, price premium, volume premium, and share premium. Finally, firm-level performance measures assess the value created by the brand to the overall corporation.

Previous research finds a positive association between brand awareness and overall customer mindset brand equity (Kim and Kum 2004; Yoo and Donthu 2001; Yoo et al.

2000), with the exception of Gil et al. (2007)'s work. These past studies generally treat brand awareness as a component of overall brand equity and suffer a few shortcomings. Some past studies consider brand awareness and brand associations as a joint dimension, causing difficulty in untangling the effect of brand awareness from brand association (e.g., Gil et al. 2007; Yoo et al. 2000). Furthermore, past studies use only survey research to explore the relationship between brand awareness and mindset brand equity, calling their external generalizability into question. In contrast, the present study uses real time-series dataset including market outcome metrics, brand equity, and marketing mix information for 11 brands of consumer-packaged goods over a period of 3 years. In addition to mindset measures of brand equity, the current research also considers market outcome measures, including revenue premium, share premium, and price premium, and explores their association with brand awareness.

Brand awareness is how consumers associate the brand with the particular product that they aim to own. Brand awareness is indispensable for the communications process to emerge i.e., a top-of-the-mind awareness (Macdonald and Sharp, 2003; Netemeyer et al., 2004). Consumer receives brand awareness via effective marketing communication channel such as television, hand phone and online advertising as it gives assurance of product quality and its credibility which helps to reduce risk in product evaluation and selection while buying a product (Aaker, 1996; Buil et al., 2013; Keller and Lehmann, 2003; Rubio et al., 2014). Brand awareness significantly impacts consumer decision making where consumers generally use it as a decision heuristic which benefits the management of customer-based brand equity (Chung et al., 2013; Huang and Sarigollu, 2011; Norazah, 2013a). Indeed, brand awareness has direct effects on brand's equity (Pouromid and Iranzadeh, 2012). Thus, the following hypothesis is postulated:

*H*₁: *Brand Awareness positively influence Brand Equity.*

The Relationship Between Store Image and Brand Equity

Store image is defined as "the way in which the store is defined in the shopper's mind partly by its functional qualities and partly by an aura of psychological attribute" (Martineau, 1958). Beristain & Zorrilla (2011) define store brand equity as "a set of
components (assets and liabilities linked to a brand) that flow into a global and subjective value associated with a brand, generating a differential response from consumers. These definitions deal with the purpose of this article suggesting the influence of store image on store brand equity.

The cue utilization theory suggests that consumers use an array of cues (e.g., price, color, brand name, brand image, store image, etc.) to assess a product's quality (Olson & Jacoby, 1972). According to Smeijn et al. (2004), consumers use these cues to form an overall evaluation that will affect their attitude toward the store as a whole and its store brands. Some recent researches have tried to examine empirically the relationship between store image and store brand equity. Collins-Dodd & Lindley (2003) have demonstrated that store image plays a role in specific store brand evaluations. Thus, in this view, they have found support for the notion that store brands can be seen as extensions of the store image, and so can contribute to store differentiation in the consumer's mind. Vahie & Paswan (2006) have investigated the impact of store image on private label brand image. They find that store atmosphere and store quality positively influence the perception of private label brand's quality. This implies that focusing on store image dimensions can boost store brand equity. Marketing literature use interchangeably the concept of store brands with terms such as 'private label brands' or 'own brands' (Smeijn et al. 2004). But some researchers distinguish between store brands that bear retailer's name, i.e., own brands, and those that don't do it, i.e., others store brands (Rao et al. 2004; Ngubo, 2011a). In this study, we use the concept of store brands as those that bear retailer's name (i.e. we focus on own brands). Because they use retailer's name, own brands operate as a signal of retailer product quality (Erden & Swait, 1998).

Studies have shown that own-name branding has different effects on customers in comparison to other-name branding strategies (Dhar & Hoch, 1997). However, in line with Beristain & Zorrilla (2012), there is a lack of empirical evidence in the relationship between store image dimensions and store brand equity. In addition, although some studies, particularly those mentioned above, have investigated the relationship between store image and brand equity, little attention has been paid to the moderating role of demographic characteristics. Thus, the following hypothesis is postulated:

*H*₂: Store Image positively influence Brand Equity.

The Relationship Between Service Quality and Brand Equity

A variety of definitions for brand equity are offered in the literature (e.g., Aaker 1991, 1996; Keller 1993). For example, Keller (1993) proposed that customer-based brand equity is the differential effect of brand knowledge on consumer response to the marketing of the brand." Brand equity has other definitions and forms, such as favorable impression, attitudinal dispositions, and behavioral predilections (Rangaswamy et al. 1993). A common problem is that brand equity may be defined differently by different people. Most scholars seem to view equity as multi-dimensional, consisting of some combination of awareness, liking, preference, meaning, repeat purchase, etc. But no single method of conceptualizing and measuring brand equity may be applicable to all brands (Kartono and Rao 2005). Since consumer response in marketing is defined in terms of consumer perceptions, preferences, and behavior arising from marketing mix activity (Hartman and Spiro 2005), brand equity in this study is conceptually defined as the overall customer response to the primary predictors of brand purchase intent and behavior.

service quality is defined as the shopper's overall perception of the superiority of services provided by a service firm. Environmental psychologists argue that a critical role of the physical environment is its ability to facilitate the goals of persons within that environment (Canter 1983; Grewal et al. 2003). Similarly, physical retailing service quality is viewed as the shopper's perception of physical elements that enable or facilitate the production of a service. As service quality is identified as a dimension of brand equity, such dimensions of service quality have the potential to contribute differentially to brand equity. More specifically, quality perceptions on both dimensions of service are likely to be positively associated with customers' behavioral intentions on a specific brand (or organization) and their likelihood of remaining a customer (Bell et al. 2005; Cronin, Brady, and Hult 2000). To the extent that both physical and service quality contributes to the improvement of brand equity, the following hypothesis is suggested:

*H*₃: *Recruitment and selection positively influence organizational performance.*

Conceptual Framework

the conceptual framework of the relationship between Brand awareness, Store image, Service quality as independent variables, and their relationship on Brand equity as the dependent variable. Brand awareness with attributes like aided awareness and top of mind awareness (Immediate brand recall) (Prachi, 2015). Store image is adducted from (Watanabe et al, 2013) with attributes of service, price, product, convenience, atmosphere, and promotion. Service quality is based on the models brought forward by (Abd-Elrahman AEH, 2018) on tangibles, reliability, responsiveness, empathy, and assurance. Brand equity with awareness, brand associations, perceived quality, brand loyalty, proprietary brand assets brought to us by (Smriti, 2014).



Figure 1 Conceptual Framework

CHAPTER III

Research Methodology

Research Design

This chapter presents and describes the approaches and techniques the researcher will use to collect data and investigate the research problem. They include such as the research design, study population, sampling techniques and procedure, data collection methods, data collection instruments, data quality control (validity and reliability), procedure of data collection, data analysis, measurement of variables and ethical consideration.

Research design was well understood as a plan for the logic of conducting a research. It stands for the advance planning of the methods to be adopted to collect the relevant data and techniques to be used in their analysis, taking into account the research objectives and staff availability, time and money (Kothari, 2004). I find this necessary because it will facilitate the smooth sailing of my research work, thus making research as efficient as possible yielding maximum information with minimal effort, time and money spending. There are many types of research design in but for this study a cross sectional design is adopted. A cross-sectional study is defined as an observational research type that analyzes data of variables collected at one given point of time across a sample population (Cherry, 2012). I decided to use the cross-sectional design for this study because of the need to answer research questions using opinions of different respondents. This will majorly take both quantitative and Qualitative approaches, as the data will be collected from a cross section of employees and over a wide sample size. The quantitative approach is to be used due to its ability to examine the relationship between the variables and predict outcomes. To a lesser extent, qualitative approach will also be used (Babbie, 2011), Qualitative research is a scientific method of observation to gather non-numerical data (Babbie, 2014). This type of research "refers to the meanings, concepts definitions, characteristics, metaphors, symbols, and description of things" and not to their "counts or measures." This will give meaning as to the operations of the Sierra Leone Banking Industry.

Sampling and Sample

The population under research is staff from various banks in Sierra Leone. For the uniqueness of this research, I have decided to incorporate at least four (4) banks in Sierra Leone; The Sierra Leone Commercial Bank (SLCB), Rokel Commercial Bank (RCB), Ecobank Sierra Leone (ESL) and the United Bank for Africa Sierra Leone (UBA-SL). The top administrators are key stakeholders and participate in the decision-making process. This population will enable the researcher to obtain the necessary data for the research. From the population; random, stratified and purposive sampling techniques were employed with the aim to achieve the desired result(s), without drawbacks on quality of work. Stratified sampling will be used in selecting respondents from the groups of managers and supervisors depending on their locations. With a simple random sample, every member of the population has an equal chance of being selected. For this reason, the researcher selected only those elements (employees) that could supply the required data.

The four organizations under study are very large offices with over 1000 working staffs in the city and provinces according to the Industry Classification Benchmark taxonomy and the HR department of the four selected banks in 2021. Therefore, it indicates that if the size of the population 1169 for a 5% precision level, where the confidence level is 95%, the sample size was 289. For the benefit of this research, respondents are all management crew staff, plus other department staff such as finance, operations, audit, marketing, HR, procurement, supervisors, IT and others. Table 1 summarize population, sample size and sampling technique of respondents amongst selected Banks in Sierra Leone. According to that, out of total 320 distributed questionnaires, 289 usable responses were received making 90% as a response rate.

	0	rganiza	ations o	f				
Types of		Respor	ndents		Populatio	Sampl	Sampling Technique	
Respondents	SLCB	RC B	ESL	UB A	n	e		
Managers	55	37	78	31	146	68	Stratified sampling	
Finance	9	7	15	5	25	9	Simple random	
Operations	31	18	35	9	63	8	Simple random	
Auditors	9	10	15	7	30	7	Simple random	
Marketing	150	111	145	178	464	75	Simple random	
Human Resource	5	5	8	4	16	8	Simple random	
Information Technology	7	5	10	5	20	10	Simple random	
Supervisors	12	6	15	5	26	8	Stratified sampling	
Procurement Staff	7	5	16	4	19	10	Simple random	
Other Staff	171	102	113	65	360	86	Stratified and simple random sampling	
Total			1		1169	289		

 Table 1 population, sample size and sampling technique of respondents amongst selected Banks in Sierra Leone

Data Collection Procedures

This research aims to examine the impact of brand awareness, store image, and service quality in promoting brand equity in the banking industry of Sierra Leone by using a questionnaire from literature reviews to test the hypotheses for this research, where employees of the banking industry of Sierra Leone formed as an analysis unit of this research. To collect credible and useful data and avoid a coercive participation, the author informed the respondents that their participation was voluntary and he explained the purpose of the research and obtained a verbal consent from them before distributing the questionnaire. This research was conducted using a questionnaire which to be considered as a suitable means for effective and accurate information. Therefore, the questionnaire method was adopted as the preferred tool to answer current research objectives.

The questionnaire was review by university professors who are having an experience and knowledge in the field of marketing and banking sector to detect mistakes or a possible source of misunderstandings and to check the accuracy and validity of the professional terms and perspicuity. Validity also checked through distributed 20 questionnaires as a pilot research to make an adjustment in proportion to the responders' abilities to answer the questions, and based on their feedback the questionnaire questions was edited to be more comprehensive and accurate. The questionnaire was designed with 5-point Likert scale ranging from Strongly Agree = 5 to Strongly Disagree = 1.

Research Materials

This research was cross-sectional and completely quantitative approach. The questionnaire contains two parts and 23 items in total: demographic information, the Brand Awareness Scale (BAS), the Store Image Scale (SIS), the Service Quality Scale (SQS) and Band Quality Scale (BQS). Table1 summaries the Cronbach's alpha for these scales.

The demographic questions have 7 items. The research also assesses some demographic variables that are presented in Part 1 of the questionnaire. The respondents will be asking about gender, age, educational level, name of bank, years of experience at their organization, position held in the Organization and marital status. The BAS will be using in this research was developed by Yoo et al. (2000) and Hong-Youl Ha et al. (2011). This scale consists of 4 items with the format of a typical Five-Point Likert Scale ranging from 1 (Strongly Disagree) to 5 (Strongly Agree) with a Cronbach's alpha score of 0.757. As

indicated by Hair, Black, Babine, and Anderson (2014) the level of Cronbach's alpha that prerequisites to achieve the steadfastness and to be a good investigation is 0.7. Along these lines, the Cronbach's alpha scores for the BAS for this exploration are reliable. The SIS will be using in this research was items developed by Low, George & Lamb, Charles. (2000). This scale consists of 4 items with the format of a typical Five-Point Likert Scale ranging from 1 (Strongly Disagree) to 5 (Strongly Agree) with a Cronbach's alpha score of 0.867. As with the BAS for this exploration are reliable. The SQS will be using in this research was developed by Buil et at, (2008), Yoo et at. (2000) and Tong and Hawley (2009). This scale consists of 4 items with the format of a typical Five-Point Likert Scale ranging from 1 (Strongly Disagree) to 5 (Strongly Agree) with a Cronbach's alpha score of 0.764. As with the BAS for this exploration are reliable. The BQS items developed by Davis (2003). This scale consists of 4 items with the format of a typical Five-Point Likert Scale ranging from 1 (Strongly Disagree) to 5 (Strongly Agree) with a Cronbach's alpha score of 0.831. As with the BAS for this exploration are reliable. The Cronbach's alpha incentive for the current examination of 16 items was determined as 0.784 which for this investigation is reliable.

Variable Name	Number of	Cronbach's
	Items	alpha
Brand Awareness	4	0.757
Store Image	4	0.867
Service Quality	4	0.764
Brand Equity	4	0.831
Total	16	0.784

Table 2 The Cronbach's Alpha for Research Variables

Data Analysis Procedures

After collecting data, SPSS v.25 software was use to analyze the data with the following steps: First, test the reliability of the scale and validity of questionnaire through Cronbach's alpha coefficient. According to (Sekaran & Bougie, 2016) Cronbach's alpha indicate that values which greater than 0.70 has a high internal consistency in measured the variables and increases reliability. Second, correlation analysis shows how variables are positively related to each other. Table 3 summarizes the correlation coefficient scale. Third, factor analysis was use to find out the principal components to identify whether the factors used in the research are able to measure the variables and whether the factors used in the questionnaire are related to the variables or not. (Hair et al., 2014) reported that exploratory factor analysis (EFA) was use to explore data and provides information on the number factors needed to represent data better. In addition, all measured or observed variables are related to each factor according to the value of the load estimation factor. The main feature of EFA is that all factors are obtained only from statistical results, not from any theory, and after the factor analysis is performed the factors can be named. In other words, EFA can be analyzed without knowing the number of factors that already present in the research or which variables that belong to which constructs.

Correlation Scale	Description
$\pm 0.90 - \pm 1.00$	Very high positive or negative correlation
$\pm 0.70 - \pm 0.89$	High positive or negative correlation
±0.69 - ±0.50	Moderate positive or negative correlation
±0.49 - ±0.30	Low positive or negative correlation
±0.29 - ±0.00	Negligible correlation

Finally, regression analysis was applying to test the hypotheses that developed to determine the effect of brand awareness, store image, and service quality in promoting brand equity in the banking industry of Sierra Leone. The hypotheses that used in this

research was tested at 95% confidence level (or 5% margin of error). The rule of decision is that if the PV less than 0.05, the null hypothesis was rejected and if it is greater than 0.05 it was accepted. A number of summary tables was prepared to demonstrate the similarities and differences of brand awareness, store image, and service quality in promoting brand equity among employees of the banking industry of Sierra Leone.

Ethical Considerations

Earlier than carrying out any look at, it is predicted that the researcher familiarizes with the recommendations of moral studies behavior. This provides the researcher with the regulations on how to research correctly to ensure the honor and well-being of the research contributors. on this take a look at, we accompanied the ethical recommendations of the banking sectors in Sierra Leone, which includes receiving a compliance letter from the university, earlier than e-mailing and dispensing the questionnaire to the members. This research was directed at considering the moral ramifications at every section of the examination cycle. This research become meant to satisfy the ethical standards of planned cooperation, making certain that participants have been no longer harm, regarding their entitlement to protection, namelessness, and self-assurance. during this research, individuals had been familiar with the importance of the exam and its motivation, and the funding in this exploration is deliberate and the facts accrued all through this research was utilized for scholastic research purposes simply and might be introduced at public/worldwide scholarly gatherings in addition to distributions and was treated with exacting secrecy.

CHAPTER IV

Research Results

Descriptive Statistics

The aim of this research is to examine the relationship between HRM practices (R&S, RS, TD, HRP and PA) and the OP in the banking industry of Sierra Leone. To achieve this goal, the researcher distributes (320) questionnaires were subjected to (289) valid questionnaire for the purposes of statistical analysis. Table 4 summarizes the distribution of the questionnaire on the research sample.

	Number	Ratio
Distributed questionnaires	320	100%
Questionnaires recovered	310	97%
Non-refunded		
questionnaires	12	3%
Non-analytical		
questionnaires	9	2%
Questionnaires under		
analysis	289	90%

Table 4 The Distribution of The Questionnaire on The Research Sample

After collecting the questionnaire from the sample, the questionnaire response scale which contains 26 items was translated to a quantitative scale by giving the answer category 5 = Strongly Agree, 4 = Agree, 3 = Neither agree nor Disagree, 2 = Disagree, 1 = Strongly Disagree. The total scores of the sample respondents for each paragraph were classified as shown in Table 5:

Likert- Scale	Classification	Description
1	1 – 1.79	Strongly Disagree
2	1.8 - 2.59	Disagree
3	2.6 - 3.39	Neither agree nor disagree
4	3.4 - 4.19	Agree
5	4.2 – 5	Strongly Agree

Table 5 The Degree of Approval of the Questionnaire Paragraphs

The researcher relied on the degree of approval of the questionnaire paragraphs according to (Idek et al., 2014) the rule specified in Table 6 that the approval for the paragraph is strongly disagree if the average mean of the paragraph between 1 - 1.79, disagree if the average mean of the paragraph falls between 1.8 - 2.59, neither agree nor disagree if the average mean of the paragraph is between 2.6 - 3.39, agree if the average mean of the paragraph between 3.4 - 4.19, and strongly agree if the average mean of the paragraph between 4.2 - 5.

Brand Awareness

Table 6 shows the mean scores for the BA. The respondents' mean scores for BA items extend from 3.56 to 3.72. Simultaneously their standard deviation exhibited that the things don't present a high deviation from the normal mean among things. Consequently, the respondents' mean scores for generally speaking BA were 3.67. These scores show that the clients view of the BA did by their association are satisfactory and agree.

Items	Means	STD	Degree of Approval
BA1	3.72	.899	Agree
BA2	3.72	1.041	Agree
BA3	3.56	1.060	Agree
BA4	3.69	1.031	Agree

Table 6 The Mean Scores for the BA Items

Store Image

Table 7 shows the mean scores for the SI. The respondents' mean scores for SI items extend from 3.37 to 4.16. Simultaneously their standard deviation exhibited that the things don't present a high deviation from the normal mean among things. Consequently, the respondents' mean scores for generally speaking SI were 3.86. These scores show that the clients view of the SI did by their association are satisfactory and agree.

Table 7 The Mean Scores for the SI Items

Items	Means	STD	Degree of Approval
SI1	3.37	1.033	Natural
SI2	4.16	1.065	Agree
SI3	3.82	1.009	Agree
SI4	4.11	.999	Agree

Service Quality

Table 8 shows the mean scores for the SQ. The respondents' mean scores for SQ items extend from 3.66 to 4.64. Simultaneously their standard deviation exhibited that the things don't present a high deviation from the normal mean among things. Consequently, the respondents' mean scores for generally speaking SQ were 4.03. These scores show that the clients view of the SQ did by their association are satisfactory and agree.

Items	Means	STD	Degree of Approval
SQ1	3.66	.860	Agree
SQ2	4.64	.678	Strongly Agree
SQ3	4.01	.924	Agree
SQ4	3.80	.856	Agree

 Table 8 The Mean Scores for the SQ Items

Brand Equity

Table 9 shows the mean scores for the BE items. The respondents' mean scores for BE items range from 2.81 to 3.22. At the same time their standard deviation demonstrated that the items do not present a high deviation from the average mean among items. Therefore, the respondents' mean scores for BE (overall) were 2.95. These scores indicate that the employees' perceptions of the BE carried out by their organization are natural.

Items	Means	STD	Degree of Approval
BE1	2.81	1.518	Natural
BE2	2.94	1.612	Natural
BQ3	3.22	1.275	Agree
BE4	2.83	1.605	Natural

 Table 9 The Mean Scores for the BE Items

Demographic Characteristics of Respondents

Demographic characteristics of respondents that have been captured in this research include 7 different aspects; gender, age, educational level, marital status, position held in the organization, duration spent with the organization and kindly select your organization. First, gender which was measured in two categories of male and female. Second, age which was measured in four categories having options of 21-30 years, from 31 - 40 years, from 41 - 50 years, and more than 50s years. Third, educational level was measured in four categories diploma, bachelor, master and PhD. Fourth, marital status which was takes in four categories of single, married, divorced and separated. Fifth, position held in the organization which was measured in five categories. Sixth, duration spent with the organization which was measured in four categories. Finally, kindly select your organization which was measured in four categories.

Gender

Gender respondents were selected in two categories: male and female. In a data from the banking industry of Sierra Leone, the majority of male and female respondents were 70% and 30% respectively. Table 10 summarizes the sample distribution by gender.

Variables	Frequency	Percent
Gender		
Male	202	70%
Female	87	30%
Total	289	100%

Table 10 Sample Distribution by Gender
--

Age

In the banking industry of Sierra Leone data, respondents were of different age groups in a relative distribution as shown in Table 11. The highest representation is respondents who belong to the 41 - 50 years by 42%. The rate of aging between 31 - 40 years was 32%, respondents over 50 years was 26%.

 Table 11 Sample Distribution by Age

Variables	Frequency	Percent
Age		
31 - 40	92	32%
41- 50	122	42%
Over 50	75	26%

Total	289	100%
-------	-----	------

Educational Level

The educational level of respondents was measured in three categories as shown below in Table 12. In the banking industry of Sierra Leone, the highest percentage of respondents who obtained bachelor is 89%, and master is 11%.

Variables	Frequency	Percent
Educational Level		
Bachelor	256	89%
Master	33	11%
Total	289	100%

 Table 12 Sample Distribution by Educational Level

Marital Status

The respondents' marital status was obtained using three categories single, married, and divorces. In the banking industry of Sierra Leone, the respondents who are married 90% and who are single 10%. Table 13 summarizes the sample distribution by marital status.

Variables	Frequency	Percent
Marital Status		
Single	29	10%
Married	260	90%
Total	289	100%

Table 13 Sample Distribution by Marital Status

Position held in the organization

The respondents' position held in the organization was obtained using ten categories' Managers 24%, Finance 3%, Operations 3%, Auditors 2%, Marketing 26%, Human Resource 3%, Information Technology 4%, Supervisors 3%, Procurement Staff 4%, Other Staff 30%. Table 14 summarizes the sample distribution by position held in the organization.

Table 14 Sample Distribution by Position held in the organization

Variables	Frequency	Percent
Position held in the organization		
Managers	68	24%
Finance	9	3%

Operations	8	3%
Auditors	7	2%
Marketing	75	26%
Human Resource	8	3%
Information Technology	10	4%
Supervisors	8	3%
Procurement Staff	10	4%
Other Staff	86	30%
Total	289	100%

Duration spent with the organization

The respondents' Duration spent with the organization was obtained using five categories. In the banking industry of Sierra Leone, the respondents who have 2-4 years 38%, who have 4-6 33%, who have over 6 years 18% and who have 1-2 11%. Table 15 summarizes the sample distribution by Duration spent with the organization.

Variables	Frequency	Percent
Duration spent with the organization		
-2 years	3	1%
-4 years	09	8%
-6 years	5	3%
ver 6 years	1	8%
otal	89	00%

 Table 15 Sample Distribution by Duration spent with the organization

Kindly select your organization

The respondents' Kindly select your organization was obtained using four categories. In the banking industry of Sierra Leone, the respondents who are from SLCB 28%, who are from UBA-SL 24%, who are from ESL 19% and who are from RCB 29%. Table 16 summarizes the sample distribution by Kindly select your organization.

Variables	Frequency	Percent
Kindly select your organization		
SLCB	81	28%
UBA-SL	69	24%
ESL	56	19%
RCB	83	29%
Total	289	100%

Table 16 Sample Distribution by Kindly select your organization

Correlation Analysis

The aftereffects of the relationship examination which demonstrates that all the four builds were emphatically connected with one another with 0.01 criticalness esteem are appeared in Table 17. The connection among BA and BE (R= 0.512, $\rho = 0.01$) is counted as a noteworthy and moderate positive relationship. The connection among SI and BE (R= 0.746, $\rho = 0.01$) is counted as a high positive relationship. The relationship coefficient among SQ and BE (R = 0.764, $\rho = 0.01$ is counted as a high positive connection.

Table 17 Corre	lation between	Variables
----------------	----------------	-----------

	BA	SI	SQ	BE
BA	1			

SI	.686**	1		
SQ	.627**	.745**	1	
BE	.512**	.746**	.764**	1
N=289	I		I	

**. Correlation is significant at the 0.01 level (2-tailed).

BA: Brand Awareness; SI: Store Image; SQ: Service Quality; BE: Brand Equity.

Exploratory Factor Analysis

Exploratory factor analysis (EFA) empowers the creator to diminish the watched factors to littler numbers and distinguish the connection between them (Hinkin, 1998). Principal components analysis (PCA) strategy following by the Promax with Kaiser Normalization pivot technique was utilized to separate the components. As proposed by Hair et al. (2014) the creator kept just those things which stacked 0.4 or above on single thing. Table 18 demonstrated KMO and Bartlett's Test and the subsequent incentive for telecommunications enterprises in Gambia was 0.745 which adequately agree to (Kaiser and Rice, 1974) of required example esteem.

Table 18KMO and Bartlett's Test

Kaiser-Meyer-Olkin Adequacy	Me	asure of Sampling	0.745
Bartlett's Test	of	Approx. Chi-Square	1756.855
Sphericity		df	120
		Sig.	0.000

The creator inspected the relationship between the factors and the visual assessment demonstrated a critical connection at p = 0.01. The EFA investigation separated four

particular variables clarifying 63.82% of the total variance. The SI had four items which indicated 0.815 to 0.889 stacking range and the build clarified 21.71% of the total variance. The BE had four items stacked between 0.791 to 0.848, and clarified 19.21% of the total variance. The BA had four items stacked from 0.699 to 0.813 and clarified 12.84% of the total variance. The SQ had four items stacked between 0.622 to 0.887 and clarified 10.06% of the total variance.

Hypotheses Testing

The analysts utilized the regression analysis for three main hypotheses:

The Relationship between Brand Awareness and Brand Equity.

Hypothesis H₁ posits that BA positively influence BE. As shown in Table 19 the linear regression analysis demonstrated that the path estimates between BA and BE was significant ($F_{(1,287)} = 102.064$, p < 0.05, $R^2 = 0.262$). Also, the model coefficient shows that BA were positive and statistically significant to BE ($T_{(287)} = 10.103$, $\beta = 0.119$, p < 0.05). According to the lower and upper bound of the 95% confidence interval if zero falls between them, then the hypothesis will be rejected. If zero falls outside of the interval, then the hypothesis will be accepted. In Table 19 shows that zero does not fall between the lower and upper bound of the 95% confidence interval (LLCI= 0.096, ULCI= 0.143), so the author infers that the effect of BA on BE is significantly different from zero. Therefore, hypothesis H₁ was accepted.

Mod	el Sum	mary											
м		R	Ad	just	Cha	nge Sta	tistic	s					
Mo	R Squar		ed	R	R Square F			161	162				
del		e	Squ	ıare	Cha	nge	Cha	ange	df1	df2	Sig. F Change		ige
1	.512 ^a	.262	.26	0	.262		102	2.064	1	287).	000	
a. Pre	edictors	: (Con	istant)	, BA									
ANO	VA ^a												
Mode	el			um quare	of es	df		Mea	n Square	F		Sig.	
	Reg	ressio	n 24	418.1	87	1		2418	8.187	102.0	54	.000 ^b	
1	Resi	dual	6	799.8	841	41 287		23.69	93				
	Total		92	218.0	028	288	88						
a. De	penden	t Vari	able: I	BE									
b. Pre	edictors	: (Cor	nstant)	, BA									
Coef	ficients	a											
						Star	ndaro	dize					
		Unstar	ndard	ized	d					95.0% Confide		Confidence	
		(Coefficients		cs Co		effici	ent	t	Sig.	In	Interval for B	
						S				8			1
		1	В	Std	. Erro	or Beta	a					ower	Upper
											B	ound	Bound
1	(Cons		29.25 3	.834	4				35.102	.000	27	7.617	30.898
	BA		119	.012	2	.512	2		10.103	.000	.0	96	.143
a. De	penden	t Vari	able: I	ЗE		•				•	•		

Table 19 I	Regression	analysis	of BA	on BE
------------	------------	----------	-------	-------

The Relationship between Store Image and Brand Equity.

Hypothesis H₂ posits that SI positively influence BE. As shown in Table 20 the linear regression analysis demonstrated that the path estimates between SI and BE was significant ($F_{(1,287)} = 360.322$, p < 0.05, $R^2 = 0.557$). Also, the model coefficient shows that SI were positive and statistically significant to BE ($T_{(287)} = 18.982$, $\beta = 0.496$, p < 0.05). According to the lower and upper bound of the 95% confidence interval if zero falls between them, then the hypothesis will be rejected. If zero falls outside of the interval, then the hypothesis will be accepted. In Table 20 shows that zero does not fall between the lower and upper bound of the 95% confidence interval (LLCI= 0.445, ULCI= 0.547), so the author infers that the effect of SI on BE is significantly different from zero. Therefore, hypothesis H₂ was accepted.

Mode	el Sumn	nary						
Mod		R	Adjusted	Change	Statistics			
el R Square			R Squa Change	are F Change	df1	df2	Sig. F Change	
1	.746 ^a	.557	.555	.557	360.32 2	1	287	.000
a. Pre	dictors:	(Constar	nt), SI				1	
ANO	VA ^a							
Mode	9]		Sum Squares	of df	Mean	Square	F	Sig.
	Regre	ession	5131.078	1	5131.0	78	360.322	.000 ^b
1	Resid	ual	4086.949	287	14.240)		
	Total		9218.028	288				
a. Dej	pendent	Variable	: BE	I			1	I
b. Pre	dictors:	(Constan	nt), SI					

 Table 20 Regression analysis of SI on SE

				Standardize				
Model		Unstandar	dized	d			95.0%	Confidence
		Coefficien	ts	Coefficient	4	C:a	Interval fo	or B
		B Std. Error		s	ι	Sig.		
				Beta			Lower	Upper
		B Std. Error		Deta			Bound	Bound
	(Constan	17.802	1.044		17.051	.000	15.747	19.856
1 t)		17.002	1.044		17.051	.000	13.747	17.050
	SI	.496	.026	.746	18.982	.000	.445	.547

The Relationship between Service Quality and Brand Equity.

Hypothesis H₃ posits that SQ positively influence BE. As shown in Table 21 the linear regression analysis demonstrated that the path estimates between SQ and BE was significant ($F_{(1,287)} = 403.388$, p < 0.05, $R^2 = 0.584$). Also, the model coefficient shows that SQ were positive and statistically significant to BE ($T_{(287)} = 20.085$, $\beta = 0.645$, p < 0.05). According to the lower and upper bound of the 95% confidence interval if zero falls between them, then the hypothesis will be rejected. If zero falls outside of the interval, then the hypothesis will be accepted. In Table 21 shows that zero does not fall between the lower and upper bound of the 95% confidence interval (LLCI= 0.582, ULCI= 0.708), so the author infers that the effect of SQ on BE is significantly different from zero. Therefore, hypothesis H₃ was accepted.

Table 21 Mediation	analysis	of SQ an	d BE
--------------------	----------	----------	------

Mode	el Summ	ary						
Mod		R	Adjusted	Change Star	tistics			
el	R		Adjusted R Square	R Square Change	F Change	df1	df2	Sig. F Change

1	.764 ^a .	584	.583	3	584	403 8		1		287	.000	
a. Pro	edictors: (C	Consta	nt), SQ	2							1	
ANC	OVA ^a											
Mod	el		Sum Squar	of	df	Me	ean Sc	luare	F		Sig.	
	Regress	ion	5386.	018	1	538	86.018	8	4	03.388	.000 ^b	
1	Residua	ıl	3832.	010	287	13.	352					
	Total		9218.	028	288							
a. De	ependent V	ariabl	e: BE		I							
b. Pr	edictors: (C	Consta	int), SQ	2								
Coef	ficients ^a											
Mod	el		tandard fficient		d	ndardize efficient	t		Sig.	In	5.0% terval fo	Confidence r B
		B Std. Error		or Be	ta					ower ound	Upper Bound	
1	(Constan t)	14.3	14	1.158			12.3	62	.000) 12	2.035	16.593
	SQ	.645		.032	.76	4	20.0	85	.000) .5	82	.708
a. De	ependent V	ariabl	e: BE	I	I		1			I		1

Overview of Hypotheses Testing

Table 22 Overview of Hypotheses Testing

	Linkage	<i>R</i> ²	Р	β	Hypotheses Acceptance
H_1	BA► BE	0.262	0.000	0.119	Accepted

H_2	SI► BE	0.557	0.000	0.496	Accepted
H3	SQ► BE	0.584	0.000	0.684	Accepted

CHAPTER V

Discussion and Conclusion

Discussion

The amount of research about the impact of brand awareness, store image, and service quality in promoting brand equity in the banking industry of Sierra Leone is limited. Accordingly, the current study aimed at classifying this issue in the literature by empirically examining the relationship between brand awareness, store image, and service quality in promoting brand equity among the employees of the banking industry of Sierra Leone.

The results showed that the relationship between BA and BE is at 26.2%. This indicated that BA have a very weak relationship but positively significant to BE. The results showed that brand awareness predominantly affects brand equity in the sense that customers' decisions to buy a product or brand depends on their awareness of the product or brand knowledge. These consumers get input and awareness of the particular product or brand in comparison to competing products or brands and know how it looks and its characteristics from the social media. Indeed, they can quickly recall the symbol or logo of the particular product or brand are consistent with preceding research (i.e. Chung et al., 2013; Huang and Sarigollu, 2011; Pouromid and Iranzadeh, 2012) where brands are more likely to be considered or chosen based on the consumers' brand awareness.

The results showed that the relationship between SI and BE is at 55.7%. This indicated that SI have a moderate relationship and positively significant to BE. This may be due to the fact that consumers who perceive a store as a competent firm, with accumulated knowhow and a capacity to innovate in order continuously to adapt the market, may also perceive that it has a greater capacity to develop high-quality own brands that satisfy their desires and needs. The research considers this contribution to be of interest given that it has not previously been studied in other works. The inter-relationship between the store and distributor brands is complex and may be addressed in terms of image, value and

loyalty, among other perspectives. This work seeks to explain the way in which store image contributes to increase store brand equity, in order to extract management implications targeted at increasing the value provided by the retailer to the market.

The results showed that the relationship between SQ and BE is at 26.2%. This indicated that SQ have moderate relationship and positively significant to BE. in view of the fact that services and products almost always co-exist, managers can benefit by making a distinction between services and products, and by acknowledging their respective antecedents and the distinctive contributions to the building of a positive reputation. It is widely agreed in all manufacturing industries, and in most service industries, that improvement in service quality will certainly contribute to a positive reputation. However, product quality should also become a priority of managers if they are to build a strong reputation for their firms. This is because customers value products as well as services in the course of accumulating a judgment over time about quality which determines the effectiveness of the reputation building activities of a firm, especially in service industries (Nguyen and Leblanc, 2001). Therefore, managers must try to distinguish the specific roles that service quality and product quality play in building a reputation and, when available resources are scarce, emphasize whichever of these has stronger influence.

Conclusion

Customers decisions to use a bank in Sierra Leone are influenced by factors such as brand recognition, store image, and the quality of the service. These findings will considerably improve marketing practitioners understanding of these factors. As a result, the purpose of this research is to better understand the primary aspects that determine how loyal clients may be generated and maintained in a highly competitive banking business in this country. Research undertaken prior to this study found that a reduced pricing for services, a positive image of the store, and a high level of customer satisfaction were critical factors in creating brand equity. According to findings from this study, a strong brand/store image plays a significant effect in the choice of banking businesses, their purchasing decisions, and customer loyalty. Customers preferred lower-priced goods and services with high value-added offerings, according to the research. If you want to succeed in business, you need to be able to provide high-value services at a reduced cost while maintaining your

competitive edge. As a result, customers are more likely to suggest the service to friends and family, as well as maintain a long-term connection with the service provider. The results obtained lead us to propose a management approach consisting of strengthening the components of the store brand equity through suitable management of the store image, in accordance with an integrated and coherent long-term strategy.

With regard to the brand associations, the company can try to insert the store image into the set of associations that make up the identity of the store brand. The store image is not limited to its marketing dimension, but may include organizational associations, closely linked to vision, mission and corporate values. The research is referring to questions such as proximity to customers, innovation, experience and success. These organizational associations, as well as transmitting credibility, allow store brands to associate themselves with the values and culture of the store, ultimately generating positive associations, which may have an important emotional character, and, which contribute to increase the equity of these brands.

In order for this image to be effectively transmitted, the relationship between the store and the store brand needs to be shown, through communication actions that highlight the close relationship between the store brand and the corporate associations it is intended to provoke. With regard to quality, this study shows that a good store image can contribute to increase the perceived quality of store brands, thus supporting efforts made by the retailer to offer good quality in its products. This is quite important, since as Corstjens and Lal (2000) have pointed out, store brands will mark the establishment out if consumers consider them to be quality brands, which leads to an increase in the costs of changing to other retailers and causes loyalty to the establishment. At the same time, this research also suggests that store image contributes to increase store brand loyalty. In this regard, store image can be seen as a reinforcement for marketing programs launched to stimulate loyalty among consumers to these brands and, ultimately, to the establishment.

In short, the store can take advantage of the effect of its image above all the components of the store brand in increasing the value provided to the market. The actions undertaken for this purpose form part of the set of actions geared towards increasing the appeal of store brands and of the store itself. brand equity in e-banking could be considered as an outcome of customer learning procedures: customers perceive service quality through interaction with online banking websites; these online service experiences then affect their trust and satisfaction toward e-banking service, which influence their wills and likelihood to develop loyalty. Finally, loyalty will lead to brand equity, which allows online banking providers to obtain superior performance and earn greater margins. As a result, this framework could be served as a useful guide for e-banking managers who seek to differentiate their services and enhance the values of their brands, for not only does it display the sequences of brand equity building events but it also discloses the relations between e-service quality and brand equity.

Recommendation

In light of the discussions, findings and the conclusions, the following recommendations are hereby coined;

1. Customers want better value and higher quality services from the banking industry; thus, it should work to improve all institutions and explore new product offerings.

2. Since service quality influences levels at which customers become loyal, the company must improve its old marketing and advertising strategies towards the new paradigm of integrated marketing communication strategies

3. If banks want to build their brand, they need to focus on providing better customer service. This is because customer loyalty is positively influenced by service quality.

4. Relationship marketing, for example, can be used by banks to increase client happiness and build customer loyalty.

Reference

- Csikszentmihalyi, M. (1975) Play and Intrinsic Rewards, Humanistic Psychology, 15(3), 1975, p. 41-63.
- Csikszentmihalyi, M. (1990) Flow: The Psychology of Optimal Experience, New York: Harpers Perennial.
- Csikszentmihalyi, M. (2004) Materialism and the evolution of consciousness, In: T. Kasser & A. D. Kanner (Eds.), Psychology and consumer culture: The struggle for a good life in a materialistic world (p. 91–106). Washington, DC: American Psychological Association.
- Csikszentmihalyi, M.; Abuhamdeh, S.; Nakamura, J. (2005) Flow. În A. Elliot & C. S. Dweck (Eds.), Handbook of competence and motivation (p. 598–608), New York: Guilford Press.
- Delle Fave, A.; Massimini, F. (2004) Bringing subjectivity into focus: Optimal experiences, life themes, and person-centered rehabilitation. In P. A. Linley & S. Joseph (Eds.), Positive psychology in practice (p. 581–597). Hoboken, NJ: Wiley & Sons.
- Delle Fave, A.; Massimini, F. (2005) The relevance of subjective well-being to social politics: Optimal experience and tailored intervention, In F. A. Huppert, N. Baylis, & B. Keverne (Eds.), The Science of well-being (p. 379–402). New York: Oxford University Press.
- Aaker, D. A. (1991). Managing Brand Equity: Capitalizing on the Value of a Brand Name. New York: Free Press.
- Aaker, D. A. (1992). The Value of Brand Equity. Journal of Business Strategy, 13(4), 27-32.
- Aaker, D. A. (1996). Building Strong Brands. New York: The Free Press.

- Keller, K. L. (2009). Building strong brands in a modern marketing communications environment. Journal of Marketing Communications, 15(2/3), 139-155.
- Verbeke, W., Vereir, I., Pieniak, Z. and Brunso, K. (2005). Consumers' Quality Perception as a Basis for Fish Market Segmentation in Belgium. Gent: Ghent University.
- Farjam, S. and Hongyi, X. (2015). Reviewing the Concept of Brand Equity and Evaluating ConsumerBased Brand Equity (CBBE) Models. International Journal of Management Science and Business Administration, 1(8), 14-29.
- Aaker D A (1992), "Managing the Most Important Assets: Brand Equity", Planning Review, Vol. 20, No. 5, pp. 56-58.
- Abd-Elrahman AEH, 2018. A Review of Telecommunications Service Quality Dimensions. Sch J Appl Sci Res, 1(1), pp. 10-18.
- Abu Muammar, F., 2005. Measurement of the quality of service in banks operating in the Gaza Strip. The Islamic University Journal (Humanities series), 13(1), pp. 105-177.
- Adam, J and Kamuzora, F., 2008. Research Methods for Business and Social Studies. s.l.: Mzumbe Book Project, Morogoro.
- Ailawadi. K.L., and Keller. K.L., 2004. Understanding retail branding: conceptual insights and re- search priorities. Journal of Retailing, Volume 80, pp. 331-342.
- Amin, 2005. Social Science Research: Conception, Methodology and Analysis. Kampala: Makerere University Press.
- Anderson J R (1983), The Architecture of Cognition, Cambridge, Harvard University Press, MA.
- Atilgan, E., Aksoy. S., & Akinci. S., 2007. Determinants of the brand equity, Marketing Intelligence and Planning. 23(3), pp. 237-248.

- Available at: https://adage.com/article/btob/top-100-b-b-advertisers-cut-spending-10-2/277835
- Azad. A. K., Al Muzahid. M., & Kamal. A., 2013. Intermediating Role of Brand Knowledge over Brand Equity: A Case Study of Export Import Bank of Bangladesh Ltd. Proceedings of 9th Asian Business Research Conference 20-21 December, BIAM Foundation, Dhaka, Bangladesh ISBN: 978-1-922069-39-9.
- Babbie, E. &. M., 2011. The Practice of Social Research. Cape Town: Oxford University press.
- Babbie, E., 2014. The Basics of Social Research (6th ed.). Belmont, California: Wadsworth Cengage.
- Balaji. M., 2011. Building strong service brands: the hierarchical relationship between brand equity dimensions. The IUP Journal of Brand Management, 8(3), pp. 7-24.
- Beristain, J.J., & Zorrilla, P. (2011). The relationship between store image and store brand equity: A conceptual framework and evidence from hypermarkets. Journal of Retailing & Consumer Services, 18(6), p. 562-574.
- Beristain. J.J., and Zorrilla. P., 2011. The relationship between store image and store brand equity: A conceptual framework and evidence from hypermarkets. Journal of Retailing and Consumer services, Volume 18, pp. 562-574.
- Bouteille, R., Cordelle, S., Laval, C., Tournier, C., Lecanu, B., This, H., & Schlich, P. (2013). Sensory Exploration of the Freshness Sensation in Plain Yoghurts and Yoghurt-like Products. Food Quality and Preference, 30, p. 282-292.
- Brodie R J and Benson-Rea M (2016), "Country of Origin Branding: An Integrative Perspective", Journal of Product & Brand Management, Vol. 25, No. 4, pp. 322-336.

- Buil, I., de Chernatony, L. and Martinez, E., 2008. A cross-cultural validation of the consumer- based brand equity scale. Journal of Product & Brand Management, 17(6).
- Buil. I., de Chernatony. L., & Martínez. E., 2013. Examining the role of advertising and sales promotions in brand equity creation. Journal of Business Research, 66(1), pp. 115-122.
- Byrne, B.M. (2009). Structural equation modeling with AMOS: basic concepts, applications, and programming (2nd ed.). New York: Routledge.
- Chellappa, R. K., Sambamurthy, V., & Saraf, N., 2010. Competing in crowded markets: Multimarket contact and the nature of competition in the enterprise systems software industry. Information Systems Research, 21(3), pp. 614-630.
- Cheng-Hsui Chen A (2001), "Using Free Association to Examine the Relationship Between the Characteristics of Brand Associations and Brand Equity", Journal of Product & Brand Management, Vol. 10, No. 7, pp. 439-451.
- Cherry, L. S., 2012. Antibiotic use for treating dental infections in children. A survey of dentists' prescribing practices. J Am Dent Assoc 2012; 143(1, pp. 31-38.
- Clayton M and Heo J (2011), "Effects of Promotional-Based Advertising on Brand Associations", Journal of Product & Brand Management, Vol. 20, No. 4, pp. 309-315.
- Collins A M and Loftus E F (1975), "A Spreading-Activation Theory of Semantic Processing", Psychological Review, Vol. 82, No. 6, p. 407.
- Collins-Dodd, C., & Lindley, T. (2003). Store brands and retail differentiation: the influence of store image and store brand attitude on store own brand perceptions. Journal of Retailing & Consumer Services, 10(6), p. 345-352.
- Cotton, B. C., & Babb, E. M. (1978). Consumer response to promotional. Journal of Marketing, 42(3), 109-113.
- Creswell, J., 2013. Research Design: Qualitative, Quantitative, and Mixed Methods Approaches. Lincoln: SAGE Publications.
- Cronbach, L. J., 1946. Response Sets and Test Validity. Educational and Psychological measurement Volume: 6 issues: 4, pp. 475-494.
- Csikszentmihalyi. M., 2000. Beyond Boredom and Anxiety; Jossey-Bass: San Francesco, CA, USA.
- Czellar, S., & Denis, J. (2002). Un modèle intégrateur du capital-client de la marque: une perspective psycho-cognitive. Recherche et Applications en marketing, 17(1), p. 42-55.
- Davis, D., 2003. The effect of brand equity in supply chain relationships, University of Tennessee dissertation, Knoxville, TN.
- De Wulf, K., Oderkerken-Shröder, G., Goedertier, F., & Van Ossel, G. (2005). Consumer perceptions of store brands versus national brands. Journal of Consumer Marketing, 22(4), p. 223-232.
- Denmark, J. R., 2010. Corporate branding: the role of vision in implementing the corporate innovative Marketing. 6(1).
- Dodou, Dimitra., 2010. Five-Point Likert Items: t Test Versus Mann–Whitney–Wilcoxon. Practical Assessment, Research and Evaluation. 15..
- Dudovskiy, J., 2018. The Ultimate Guide to Writing a Dissertation in Business Studies: a step-by-step assistance. s.l.:Dudovskiy, John.
- Faircloth. J.B., Capella. L.M., & Alford. B.L., 2001. The effect of brand attitude and brand image on brand equity. J Mark Theory Pract. 9(3), p. 61–75.
- Faris, P. W., Bendle, N. T., Pfeifer, P. E., & Reibstein, D. J , 2010. Marketing Metrics: The definitive guide to measuring marketing performance.NJ: Pearson, Upper Saddle.

- Fatma M, Khan I and Rahman Z (2016), "How Does Corporate Association Influence Consumer Brand Loyalty? Mediating Role of Brand Identification", Journal of Product & Brand Management, Vol. 25, No. 7, pp. 629-641.
- Gefan, D. (2002). Customer loyalty in e-commerce, Journal of the Association of Information System, Vol. 3, No. 1, pp. 27-51.
- Grönroos. C., 1984. A service quality model and its marketing implications. European Journal Marketing, Volume 14, pp. 30-44.
- Hair, J. F., Black, W. C., Babin, B. J., & Anderson, R. E. (2014). Multivariate Data Analysis (7th ed.). Pearson Education Limited.
- Hal Dean D (2004), "Evaluating Potential Brand Associations through Conjoint Analysis and Market Simulation", Journal of Product & Brand Management, Vol. 13, No. 7, pp. 506-513.
- Henderson G R, Iacobucci D and Calder B J (1998), "Brand Diagnostics: Mapping Branding Effects Using Consumer Associative Networks", European Journal of Operational Research, Vol. 111, No. 2, pp. 306-327.
- Hinson, R., Owusu-Frimpong, N., & Dasah, J., 2009. Key motivations for bank patronage in Ghana. International Journal of Bank Marketing, 27(5), pp. 381-399.
- Hirut, S., 2015. The Effect of Service Quality on Customer Based Brand Equity (A Case of Commercial Bank of Ethiopia). A Thesis Submitted to Addisababa University in Partial Fulfillment of The Requirement for The Masters of Business Administration (Mba) In Management.
- Hong-Youl Ha, Joby John, Swinder Janda, Siva Muthaly, 2011. The effects of advertising spending on brand loyalty in services. European Journal of Marketing, 45(4).
- Hunjira, A. e. a., 2011. Relationship between customer-based brand equity and service quality of Islamic banks. Journal of World applied sciences, 13(3), pp. 453-459.

- Idek, S., Fong, L. L., Sidhu, G. K., & Hoon, T. S. (2014). The Use of Consciousnessraising Tasks in Promoting the Correct Use of the Verb "Be" among Students in Vocational Colleges. Theory and Practice in Language Studies, 4(9), 1792–1803.
- James D (2005), "Guilty through Association: Brand Association Transfer to Brand Alliances", Journal of Consumer Marketing, Vol. 22, No. 1, pp. 14-24.
- Jeon J O and Baeck S (2016), "What Drives Consumer's Responses to Brand Crisis? The Moderating Roles of Brand Associations and Brand-Customer Relationship Strength", Journal of Product & Brand Management, Vol. 25, No. 6, pp. 550-567.
- Jevons C (2005) "Names, Brands, Branding: Beyond the Signs, Symbols, Products and Services", Journal of Product & Brand Management, Vol. 14, No. 2, pp. 117-118.
- John D R, Loken B, Kim K and Monga A B (2006), "Brand Concept Maps: A Methodology for Identifying Brand Association Networks", Journal of Marketing Research, Vol. 43, No.4, pp. 549-563.
- Kahneman, D., 2012. Pułapki myślenia. O myśleniu szybkim i wolnym [The trap of thinking. Thinking fast and slow]. Media Rodzina: Poznań.
- Kakati. R.P, & Choudhury. S., 2013. Measuring customer-based brand equity through brand building blocks for durables. Journal of Brand Management, 10(2), pp. 24-41.
- Kankiriho, K., 2014. Techniques for Quality Data analysis. Kampala: s.n.
- Kay. M.J., 2006. Strong brands and corporate corporate. Eur J Mark. 40(7/8), p. 742–760.
- Keller K L (1993), "Conceptualizing, Measuring, and Managing Customer-Based Brand Equity", The Journal of Marketing, Vol. 57, No. 1, pp. 1-22.
- Keller. K.L., 1993. Conceptualizing, measuring, and managing customer-based brand equity. Journal of Marketing, Volume 57, p. 1–22.
- Kothari, C., 2004. Research Methodology: Methods and Techniques. New Delhi.

- Kreuzbauer. R., & Malter. A.J., 2005. Embodied cognition and new product design: changing product form to influence brand categorization. Journal of Product Innovation Management, 22(2), p. 165–176.
- Lambert. J. Csikszentmihalyi, M., 2019. Facilitating or foiling flow: The role of momentary perceptions of feedback. J. Posit. Psychol., pp. 1-12.
- Lim. W.M., 2014. Understanding the influence of online flow elements on hedonic and utilitarian online shopping experiences: A case of online group buying. Journal of Information System, Volume 28, pp. 287-306.
- Low G S and Lamb Jr. C W (2000), "The Measurement and Dimensionality of Brand Associations", Journal of Product & Brand Management, Vol. 9, No. 6, pp. 350-370.
- Low, George & Lamb, Charles., 2000. The Measurement and Dimensionality of Brand Associations. Journal of Product & Brand Management, Volume 9.
- Maddox, K., 2009. Top 100 b-to-b advertisers cut spending 10.2%. Advertising Age. [Online]
- Moser, G. and Uzzell, D.L. (2003), "Environmental psychology", in Millon, T., Lerner,M.J. and Weiner, I.B. (Eds), Handbook of Psychology: Personality and SocialPsychology, John Wiley and Sons Inc., New York, NY.
- Ms. Michele. L, N., Dr. Grafoute. A., Dr. Zhiliang. W., & Pr. Igor-Mathieu. G., 2016. An Applied Study on Brand Equity: Factors Affecting Brand Equity Case Study of Mtn Telecom Cameroon. European Journal of Business and Management, 8(9).
- Mugenda, A., 2008. Social Science Research: Theory and Principles. Nairobi: Acts Press.
- Netemeyer, R.G., Krishan, B., Chris, P., Wang, G., Yagci, M., Dean, D., Ricks, J. and Wirdh, F. (2004), "Developing and validating measure of facets of customer-based brand equity", Journal of Business Research, Vol. 57 No. 1, pp. 209-224.

- Norazah, M.S. (2013a), "Green awareness effects on consumer's purchasing decision: some insights from Malaysia", International Journal of Asia Pacific Studies, Vol. 9 No. 2, pp. 49-63.
- Norazah, M.S. (2013b), "Young consumer ecological behaviour: the effects of environmental knowledge, healthy food, and healthy way of life with the moderation of gender and age", Management of Environmental Quality: An International Journal, Vol. 24 No. 6, pp. 726-737.
- Norazah, M.S. and Norbayah, M.S. (2013), "Dependency on smartphones: an analysis of structural equation modelling", Jurnal Teknologi (Social Sciences), Vol. 62 No. 1, pp. 49-55.
- O'Cass A and Frost H (2002), "Status Brands: Examining the Effects of NonProduct-Related Brand Associations on Status and Conspicuous Consumption", Journal of Product & Brand Management, Vol. 11, No. 2, pp. 67-88.
- O'Guinn, T. C., Allen, C. T., & Semenik, R. J., 2009. Advertising and integrated brand promotion (5th ed.). Mason, OH: South-Western.
- Osselaer, V. and Janiszewski, C. (2001), "Two ways of learning brand association", Journal of Consumer Research, Vol. 28 No. 2, pp. 202-223.
- Papasolomou, I., & Vrontis, D., 2006. Building corporate branding through internal marketing: the case of the UK retail bank industry. Journal of product & brand management, 15(1), pp. 37-47.
- Parasuraman, A., Zeithaml, V., & Berry, L. (1991). Refinement and reassessment of the SERVQUAL scale. Journal of Retailing, Vol. 67, No. 4, pp. 420-50.
- Parasuraman, A., Zeithaml, V., & Berry, L. (1988). SERVQUAL: a multiple-item scale for measuring consumer perceptions of service quality. Journal of Retailing, Vol. 64, No. 1, pp. 12-40.

- Parasuraman. (1990). An Empirical Examination of Relationships in an Extended Service Quality Model. Cambridge, MA: Marketing Science Institution.
- Park, C.W., Jaworski, B.J. and MacInnis, D.J. (1986), "Strategic brand concept-image management", Journal of Marketing, Vol. 50 No. 4, pp. 135-145.
- Pars, S. R., & Gulsel, C., 2011. The effects of brand image on consumers' choice. International Journal of Business and Social Science, 20(2), pp. 227-238.
- Peterson, L. and Low, B. (2011), "Student attitudes towards mobile library services for smartphone", Journal of Library Hi Tech, Vol. 29 No. 3, pp. 412-423.
- Pouromid, B. and Iranzadeh, S. (2012), "The evaluation of the factors effects on the brand equity of Pars Khazar household appliances based on the vision of female consumer", Middle-East Journal of Scientific Research, Vol. 12 No. 8, pp. 1050-1055.
- PWC/Ghana Bankers Association, 2014. The future of banking in Ghana, what's next? Ghana Banking Survey Report 2014, Accra: PWC/Ghana Bankers Association.
- Raaijmakers J G and Shiffrin R M (1981), "Search of Associative Memory", Psychological Review, Vol. 88, No. 2, p. 93.
- Ratcliff R and McKoon G (1988), "A Retrieval Theory of Priming in Memory", Psychological Review, Vol. 95, No. 3, p. 385.
- Rojas-Méndez J (2013), "The Nation Brand Molecule", Journal of Product & Brand Management, Vol. 22, No. 7, pp. 462-472.
- Romaniuk, J., Wight, S., & Faulkner, M., 2017. Brand awareness: Revisiting an old metric for a new world. Journal of Product & Brand Management, 26(5), pp. 469-476.
- Santos. J., 2003. E-service quality: a model of virtual service quality dimensions. Managing Ser Quality, Volume 13, pp. 233-246.

- Scott, R., & Bennett, E., 2014. Branding resources: extractive communities, industrial brands capes and themed environments. Work, Employment & Society, 29(2), pp. 278-294.
- Sekaran, U., & Bougie, R. (2016). Research Methods for Business: A Skill-Building Approach (7th ed.). Wiley & Sons Ltd.
- Shernoff. D.J., Sinha. S., Bressler. D.M., & Schultz. D., 2017. Teacher perceptions of their curricular and pedagogical shifts: Outcomes of a project-based model of teacher professional development in the next generation science standards. Front. Psychol. Volume 8, p. 989.
- Smith. G., 2004. Brand image transfer through sponsorship: a consumer learning perspective. Journal of Marketing Management, 20(3-4), p. 457–474.
- Smriti. C., 2014. Brand: 5 Main Elements of Brand Equity Explained.
- Srull T K and Wyer R S (1989), "Person Memory and Judgment", Psychological Review, Vol. 96, No. 1, p. 58.
- Srull. T.K., & Wyer. R.S., 1989. Person memory and judgment. Psychol Rev. 96(1), p. 58–83.
- Sulieman, A., 2013. Basic dimensions of the (servqual model) and its impact on the level of customer satisfaction: an empirical study of the housing bank in karak. European Scientific Journal, 9(1), pp. 21-34.

The World Bank, 2018. Sierra Leone Financial Inclusion Project (P166601). s.l.: s.n.

Thomson. M., MacInnis. D.J., & Park. C.W., 2005. The ties that bind: measuring the strength of consumers' emotional attachments to brands. Journal of Consumer Psychology, 15(1), p. 77–91.

- Till B D, Baack D and Waterman B (2011), "Strategic Brand Association Maps: Developing Brand Insight", Journal of Product & Brand Management, Vol. 20, No. 2, pp. 92-100.
- Tong, X. and Hawley, J.M., 2009. Measuring customer-based brand equity: empirical evidence from the sportswear market in China. Journal of Product & Brand Management, 18(4).
- Torelli, C. J., 2013. Globalization, culture, and branding: How to leverage cultural equity for building iconic brands in the era of globalization. New York: Palgrave Macmillan.
- Uggla H (2006), "The Corporate Brand Association Base: A Conceptual Model for the Creation of Inclusive Brand Architecture", European Journal of Marketing, Vol. 40, Nos. 7/8, pp. 785-802.
- Valavi. P., 2014. Factors influencing mobile services adoption: a brand- equity perspective. International Journal of Research In Social Sciences, 4(3), pp. 1-18.
- Vrontis, D., & Thrassou, A., 2006. Situation analysis and strategic planning: an empirical case study in the UK beverage industry. Innovative Marketing, 2(2), pp. 134-151.
- Watanabe, Eluiza & Lima-Filho, Dario & Torres, Cláudio., 2013. Store Image Attributes and Customer Satisfaction in Supermarkets in Campo Grande-MS. Revista Brasileira de Marketing. Volume 12, pp. 85-107.
- Wu C and Yen Y C (2007), "How the Strength of Parent Brand Associations Influence the Interaction Effects of Brand Breadth and Product Similarity with Brand Extension Evaluations", Journal of Product & Brand Management, Vol. 16, No. 5, pp. 334-341.
- Yoo, B., Donthu, N., & Lee, S, 2000. An examination of Selected Marketing Mix Elements and Brand Equity. Journal of the Academy of Marketing Science, 28(2).

- Zhang, 2015. The Impact of Brand Image on Consumer Behavior: A Literature Review. Open Journal of Business and Management, Volume 3, pp. 58-62.
- Zhang, X., Huang, J., Qiu, H., & Huang, Z., 2010. A consumer segmentation study with regards to genetically modified food in urban China. Food policy, 35(5), pp. 456-462.
- Zhang, Y., 2015. The impact of brand image on consumer behavior: A literature re-view. Open Journal of Business and Management, 3(1), pp. 58-60.

Appendix

QUESTIONNAIRE

Dear Respondent,

I am **Abubakarr Alex Sesay**, a student of Near East University studying a Master's Degree in MBA I am carrying out academic research which intends to investigate the impact of brand awareness, store image, and service quality in promoting brand equity in Banking Industry of Sierra Leone. This questionnaire is designed to facilitate a study on the above subject with special focus on brand equity.

You have been sampled to participate in this study and the information you give will be used strictly for academic purposes and will never be used against you or your office. The information obtained from you will be kept highly confidential. You are also requested not to write your name on this questionnaire. Fill out the questionnaire and return to me.

Thank you for your cooperation.

INSTRUCTIONS:

Please answer by ticking the appropriate answers where necessary.

SECTION A: BACKGROUND INFORMATION

Please tick your selected option

1. Gender

Male	Female

2. Age Group

21 – 30 yrs	31 – 40 yrs	41 – 50 yrs	Over 50 yrs

3. Marital status

Single	Married	Divorced	Separated

4. Highest academic qualification attained

Diploma	Bachelors	Masters	PhD

5. Position held in the Organization

Managers	
Finance	
Operations	

Auditors	
Marketing	
Human Resource	
Information Technology	
Supervisors	
Procurement Staff	
Other Staff	

6. Duration spent with the Organization

Less year	than	1	1-2 years	2-4 years	4 -6 years	Over 6 years

7. Kindly select your organization

The Sierra Leone	United Bank for Africa	Ecobank		Ecobank		Ecobank		Ecobank		Ecobank		Rokel	Commercial
Commercial	Sierra Leone (UBA-SL)	Sierra Leone		Sierra Leone		Sierra Leone Bank (I		Bank (R	CB)				
Bank (SLCB)		(ESL)											

Instructions from question - tick the number that best indicates your opinion on the questions using the following scale.

Scale	1	2	3	4	5		
	Strongly disagree	Disagree	Not sure	Agree	Strongly agree		

SECTION B: STUDY VARIABLES

	Please indicate the rate at which you rank Brand Awareness, Store Image, and Service Quality in promoting Brand Equity in your organization. Tick the scale	Strongly 1	disagree	Disagree	Sof Sure	4 Agree	c Strongly
	Brand Awareness						
8.	I believe our clients and staff can undoubtedly review this brand during any buying power						
9.	Both customers and staff have sufficient information about our brand						
10.	My Position of Choice Set is high toward the brand						

11. I believe our customers and staff can recall of competitive brands easily	
---	--

	Store Image			
12.	Regular calls from the bank are a sign of our responsiveness to customers			
13.	Our Bank has nice outlay that makes it deliver quality service to customers			
14.	Our Bank has reliable financial services for our customers			
15.	Our bank plays no game with customers account which gives them much assurance			
	Service Quality			
16.	Our customers will purchase brand with product performance and value-added services/features			
17.	I believe our customers will do financial transactions with our brand because of our high service quality			
18.	I believe Quality perception influences customers brand decision			
19.	I think innovation is key for customers choice for financial services provider			

	Brand Equity			
20	Our customers are motivated to recommend the brand that they use to other customers and none customers			
21	Our customers are not motivated to purchase alternative packages			
22	I believe customers and staff would continue to use this brand even at high usage rate			
23	I don't think our customers would be motivated to switch to another competitive brand			

THANKS FOR YOUR PARTICIPATION

Turnitin Similarity Report

THE IMPACT OF BRAND AWARENESS, STORE IMAGE, AND SERVICE QUALITY IN PROMOTING BRAND EQUITY IN THE BANKING INDUSTRY OF SIERRA LEONE

by Abu Bakarr Alex Sesay 20192482

Submission date: 18-Aug-2022 10:49AM (UTC+0300) Submission ID: 1883866320 File name: Test.docx (163.05K) Word count: 9357 Character count: 51919

THE IMPACT OF BRAND AWARENESS, STORE IMAGE, AND SERVICE QUALITY IN PROMOTING BRAND EQUITY IN THE BANKING INDUSTRY OF SIERRA LEONE

ORIGINALITY REPORT									
	4% ITY INDEX	14%	3% PUBLICATIONS	% STUDENT PAPERS					
PRIMARY S	OURCES								
	ugspace	e.ug.edu.gh		5%					
	umispac	ce.umi.ac.ug		3%					
	WWW.SC	ribd.com		2%					
	reposito	ory.smuc.edu.et		1 %					
	utamu.a			1 %					

ETHICAL COMMITTEE APPROVAL



BİLİMSEL ARAŞTIRMALAR ETİK KURULU

07.10.2021

Dear Abubakarr Alex Sesay

Your application titled **"The Impact of Brand Awareness, Store Image, and Service Quality in Promoting Brand Equity in The Banking Industry of Sierra Leone"** with the application number NEU/SS/2021/1078 has been evaluated by the Scientific Research Ethics Committee and granted approval. You can start your research on the condition that you will abide by the information provided in your application form.

Assoc. Prof. Dr. Direnç Kanol

Rapporteur of the Scientific Research Ethics Committee

Direnc Kanol

Note: If you need to provide an official letter to an institution with the signature of the Head of NEU Scientific Research Ethics Committee, please apply to the secretariat of the ethics committee by showing this document.